THIS DOCUMENT IS IMPORTANT AND REQUIRES YOUR IMMEDIATE ATTENTION. If you are in any doubt about the contents of this Document or as to the action you should take, you are recommended to seek advice from your solicitor, accountant or other independent financial adviser authorised under the Financial Services and Markets Act 2000 (as amended) if you are in the United Kingdom or, if not, by another appropriately authorised independent financial adviser.

If you sell or transfer or have sold or transferred all of your Ordinary Shares, please forward this Document, but not the personalised Form of Proxy, as soon as possible to the purchaser or transferee or to the bank, stockbroker or other agent through or to whom the sale or transfer was effected for onward transmission to the purchaser or transferee. If you sell or have sold or otherwise transferred only part of your holding of Ordinary Shares, please consult the bank, stockbroker or other agent through whom the sale or transfer was effected as to the action you should take.

# **Redx Pharma plc**

(Incorporated in England and Wales under the Companies Act 2006 with registered number 7368089)

# Proposed Voluntary Cancellation of Admission of Ordinary Shares to Trading on AIM

# **Re-Registration as a Private Limited Company**

# Adoption of New Articles of Association

and

# **Notice of General Meeting**

The Directors, whose names appear in Part I of this Document, accept responsibility, collectively and individually, for the information contained in this Document. To the best of the knowledge and belief of each of the Directors (who have all taken reasonable care to ensure that such is the case), the information contained in this Document is in accordance with the facts and does not omit anything likely to affect the import of such information.

This Document should be read in its entirety. Your attention is drawn to the letter from the Chair of the Company set out in Part I of this Document which includes a recommendation of the Directors that you vote in favour of the Resolutions to be proposed at the General Meeting referred to below.

A notice convening the General Meeting, to be held at the offices of Cooley (UK) LLP, 22 Bishopsgate, London EC2N 4BQ at 11:30 a.m. on Friday 19 April 2024 is set out at the end of this Document.

A Form of Proxy is enclosed for use by Shareholders in connection with the General Meeting. To be valid, Forms of Proxy, completed in accordance with the instructions printed thereon, must be received by the Company's registrars, Equiniti Limited, at Aspect House, Spencer Road, Lancing, West Sussex, BN99 6DA, as soon as possible but in any event by no later than 11:30 a.m. on Wednesday 17 April 2024. Shareholders who hold their Ordinary Shares in uncertificated form in CREST may alternatively use the CREST proxy voting service in accordance with the procedures set out in the CREST Manual as explained in the notes accompanying the Notice of General Meeting at the end of this Document. Proxies submitted via CREST must be received by the Company's registrars, Equiniti Limited, by no later than 11:30 a.m. on Wednesday 17 April 2024.

The appointment of a proxy will not preclude Shareholders from attending and voting at the General Meeting in person should they so wish.

# Forward-Looking Statements

This Document contains "forward-looking statements" which includes all statements other than statements of historical fact including, without limitation those regarding the Company's financial position, business strategy, plans and objectives of management for future operations, or any statements preceded by, followed by or that include the words "targets", "believes", "expects", "aims", "intends", "will", "may", "might", "anticipates", "would", "could", "shall", "estimate", "plans", "predicts", "continues", "assumes", "positioned", or similar expressions or negatives thereof. Such forward-looking

statements involve known and unknown risks, uncertainties and other important factors beyond the Company's control that could cause the actual results, performance or achievements of the Company to be materially different from future results, performance or achievements expressed or implied by such forward-looking statements. Such forward-looking statements are based on numerous assumptions regarding the Company's present and future business strategies and the environment in which the Group will operate in the future. These forward-looking statements speak only as at the date of this Document. The Company expressly disclaims any obligation or undertaking to disseminate any updates or revisions to forward-looking statements contained herein to reflect any change in the Company's expectations with regard thereto or any change in events, conditions or circumstances on which any such statements are based unless required to do so by applicable law or the AIM Rules, UK MAR or the Disclosure Guidance and Transparency Rules of the FCA. All subsequent oral or written forward-looking statements attributed to the Company or any persons acting on its behalf are expressly qualified in their entirety by the cautionary statement above.

# CONTENTS

# Page

EXPECTED TIMETABLE OF PRINCIPAL EVENTS	1
DIRECTORS, SECRETARY AND ADVISERS	2
DEFINITIONS	3
PART I LETTER FROM THE CHAIR OF REDX PHARMA PLC	6
PART II PRINCIPAL EFFECT OF RE-REGISTRATION AND ADOPTION OF NEW ARTICLES ON SHAREHOLDERS	13
PART III THE TAKEOVER CODE	15
NOTICE OF GENERAL MEETING	19

# EXPECTED TIMETABLE OF PRINCIPAL EVENTS

Event	<i>Time and/or date<sup>(1)(2)</sup></i>
Publication and posting of this Document	2 April 2024
Latest time for receipt of proxy appointments in respect of the General Meeting	11:30 a.m. on 17 April 2024
General Meeting	11:30 a.m. on 19 April 2024
Last day of dealings in Ordinary Shares on AIM	30 April 2024
Cancellation of admission of the Ordinary Shares to trading on AIM	7:00 a.m. on 1 May 2024
Matched Bargain Facility for Ordinary Shares commences	1 May 2024
Expected re-registration as a private company	week commencing 20 May 2024

#### Notes:

(1) All of the times referred to in this Document refer to London time, unless otherwise stated.

(2) Each of the times and dates in the above timetable is subject to change. If any of the above times and/or dates change, the revised times and dates will be notified to Shareholders by an announcement through a Regulatory Information Service.

# DIRECTORS, SECRETARY AND ADVISERS

Directors	Dr Jane Griffiths <i>(Chair)</i> Lisa Anson <i>(Chief Executive Officer)</i> Dr Joseph Anderson <i>(Non-Executive Director)</i> Natalie Berner <i>(Non-Executive Director)</i> Dr Bernhard Kirschbaum <i>(Non-Executive Director)</i> Peter Presland <i>(Non-Executive Director)</i> Dr Rob Scott <i>(Non-Executive Director)</i>
Company Secretary and General Counsel	Claire Solk
Registered office	Block 33, Mereside Alderley Park Macclesfield SK10 4TG
Nominated Adviser to the Company	Spark Advisory Partners Limited 5 St. John's Lane London EC1M 4BH
Joint Broker	Panmure Gordon (UK) Limited 40 Gracechurch Street London EC3V 0BT
Joint Broker	WG Partners LLP 85 Gresham Street London EC2V 7NQ
Legal advisers to the Company	Cooley (UK) LLP 22 Bishopsgate London EC2N 4BQ
Registrars	Equiniti Limited Aspect House Spencer Road Lancing West Sussex BN99 6DA

# DEFINITIONS

The following definitions apply throughout this Document, unless the context requires otherwise:

"AIM"	AIM, the market operated by the London Stock Exchange;
"AIM Rules"	the rules and guidance for companies whose shares are admitted to trading on AIM entitled "AIM Rules for Companies" published by the London Stock Exchange, as amended from time to time;
"Business Day"	a day (excluding Saturdays, Sundays and public holidays in England and Wales) on which banks are generally open for the transaction of normal banking business in London and the London Stock Exchange is open for trading;
"Cancellation"	the cancellation of admission of the Ordinary Shares to trading on AIM in accordance with Rule 41 of the AIM Rules, subject to passing of the Cancellation Resolution;
"Cancellation Resolution"	Resolution 1 to be proposed at the General Meeting;
"Companies Act"	the Companies Act 2006 (as amended from time to time);
"Company" or "Redx"	Redx Pharma plc, a company incorporated in England and Wales with company number 7368089 whose registered office is at Block 33, Mereside, Alderley Park, Macclesfield SK10 4TG;
"CREST"	the relevant system (as defined in the CREST Regulations) for the paperless settlement of trades and the holding of uncertificated securities, operated by Euroclear, in accordance with the same regulations;
"CREST Manual"	the rules governing the operation of CREST, as published by Euroclear;
"CREST member"	a person who has been admitted by Euroclear as a system- member (as defined in the CREST Regulations);
"CREST Participant ID"	the identification code or membership number used in CREST to identify a particular CREST member or other CREST participant;
"CREST participant"	a person who is, in relation to CREST, a system participant (as defined in the CREST Regulations);
"CREST personal member"	a CREST member admitted to CREST as a personal member;
"CREST Regulations"	the Uncertificated Securities Regulations 2001 (SI2001/3755), (as amended from time to time);
"CREST sponsor"	a CREST participant admitted to CREST as a CREST sponsor;
"CREST sponsored member"	a CREST member admitted to CREST as a sponsored member;
"Current Articles"	the articles of association of the Company in force as at the date of this Document;
"Directors" or "Board"	the directors of the Company, whose names are set out in Part I of this Document;
"Disclosure Guidance and Transparency Rules"	the disclosure rules and transparency rules made by the FCA pursuant to section 73A of FSMA;

"Document"	this document, containing information regarding the
	Cancellation, the Re-registration, the adoption of the New Articles and the General Meeting;
"Euroclear"	Euroclear UK & International Limited, the operator of CREST;
"Existing Ordinary Shares"	the 388,985,916 existing Ordinary Shares in the capital of the Company as 28 March 2024, being the latest practicable date prior to the publication of this Document;
"FCA"	the Financial Conduct Authority;
"Form of Proxy"	the form of proxy for use by Shareholders in connection with the General Meeting which accompanies this Document;
"FSMA"	the Financial Services and Markets Act 2000 (as amended from time to time);
"General Meeting"	the general meeting of the Company convened for 11:30 a.m. on 19 April 2024 and any adjournment thereof, notice of which is set out at the end of this Document;
"Group"	together, the Company and its subsidiary undertakings (as such term is defined in section 1162 of the Companies Act) from time to time and " <b>Group Company</b> " shall mean the Company and any such subsidiary undertaking;
"London Stock Exchange"	London Stock Exchange plc;
"Matched Bargain Facility"	the unregulated matched bargain trading facility operated by J P Jenkins for the trading of Ordinary Shares following the Cancellation;
"New Articles"	the new articles of association of the Company proposed to be adopted pursuant to Resolution 2 to be proposed at the General Meeting, a copy of which is set out in Appendix 1 to this Document and can be viewed at www.redxpharma.com;
"Notice of General Meeting" or "Notice"	the notice of the General Meeting which is set out at the end of this Document;
"Ordinary Shares"	the ordinary shares of £0.01 each in the capital of the Company;
"Panel"	the Panel on Takeovers and Mergers;
"RedCo II"	RedCo II Master Fund, L.P., a fund advised and managed by Redmile;
"Redmile"	Redmile Group, LLC;
"Registrars"	Equiniti Limited of Aspect House, Spencer Road, Lancing, West Sussex, BN99 6DA;
"Regulatory Information Service"	has the meaning given to it in the AIM Rules for any of the services approved by the London Stock Exchange for the distribution of AIM announcements;
"Re-registration"	the proposed re-registration of the Company as a private limited company;
"Re-registration Resolution"	Resolution 2 to be proposed at the General Meeting;
"Resolutions"	the resolutions to be proposed at the General Meeting;
"RM3"	RM Special Holdings 3, LLC an entity owned directly or indirectly by funds advised or managed by Redmile;

"Shareholders"	holders of Ordinary Shares from time to time;	
"Sofinnova"	Sofinnova Crossover I SLP;	
"Spark"	Spark Advisory Partners Limited, the Company's nominated adviser;	
"Takeover Code"	the City Code on Takeovers and Mergers;	
"UK MAR"	Regulation (EU) (No 596/2014) of the European Parliament and of the Council of 16 April 2014 on market abuse as it forms part of domestic law in the United Kingdom by virtue of the European Union (Withdrawal) Act 2018; and	
"United Kingdom" or "UK"	the United Kingdom of Great Britain and Northern Ireland.	

A reference to  $\ensuremath{\mathtt{\pounds}}$  is to pounds sterling, being the lawful currency of the UK.

# PART I

# LETTER FROM THE CHAIR OF REDX PHARMA PLC

(Incorporated in England and Wales with Registered No. 7368089)

Directors:

Dr Jane Griffiths (Chair) Lisa Anson (Chief Executive Officer) Dr Joseph Anderson (Non-Executive Director) Natalie Berner (Non-Executive Director) Dr Bernhard Kirschbaum (Non-Executive Director) Peter Presland (Non-Executive Director) Dr Rob Scott (Non-Executive Director) Registered Office:

Block 33, Mereside Alderley Park Macclesfield SK10 4TG

2 April 2024

# Proposed Voluntary Cancellation of Admission of Ordinary Shares to Trading on AIM Re-registration as a Private Limited Company Adoption of New Articles of Association and

# Notice of General Meeting

#### 1. Introduction

As announced by the Company earlier today, the Directors have, after an extensive review, concluded that, for the reasons set out in paragraph 2 below, it is in the best interests of the Company and its Shareholders to seek Shareholders' approval for cancellation of the admission of the Ordinary Shares to trading on AIM and for the Company to be re-registered as a private limited company. In accordance with Rule 41 of the AIM Rules, the Company has notified the London Stock Exchange of the date of the proposed Cancellation.

The Company is seeking Shareholders' approval for the Cancellation and Re-registration at the General Meeting, which has been convened for 11:30 a.m. on Friday 19 April 2024 at the offices of Cooley (UK) LLP, 22 Bishopsgate, London EC2N 4BQ. The Company is also seeking Shareholders' approval at the General Meeting for the adoption of the New Articles.

If the Cancellation Resolution is passed at the General Meeting, it is anticipated that the Cancellation will become effective at 7:00 a.m. on Wednesday 1 May 2024. The Cancellation Resolution is conditional, pursuant to Rule 41 of the AIM Rules, upon the approval of Shareholders holding not less than 75 per cent. of the votes cast by Shareholders (whether present in person or by proxy) at the General Meeting, notice of which is set out at the end of this Document.

The purpose of this Document is to seek Shareholders' approval for the Resolutions, to provide information on the background and reasons for the proposed Cancellation and the Reregistration, to explain the consequences of the Resolutions and provide reasons why the Directors unanimously consider the Resolutions to be in the best interests of the Company and its Shareholders as a whole.

The Notice of the General Meeting is set out at the end of this Document.

#### 2. Background to and reasons for the Cancellation and Re-registration

The Board has extensively reviewed and evaluated the benefits and drawbacks for the Company and its Shareholders in retaining the admission to trading of the Ordinary Shares on AIM. The Board has taken into consideration numerous factors, both positive and negative, and considered the interests of all Shareholders in reaching its decision. Following this review, the Board has concluded that the continued admission to trading of the Ordinary Shares on AIM is not appropriate and, accordingly, the Cancellation and Re-registration are in the best interests of the Company and its Shareholders as a whole for the reasons set out below.

• Limited liquidity in the Ordinary Shares and high share price volatility: There continues to be limited and inconsistent liquidity in the Ordinary Shares, as a result of which small trades in the Ordinary Shares can have a significant impact on price and, therefore, on the market valuation of the Company. The Board believes that this, in turn, has a materially adverse

impact on the Company's ability to seek appropriate financing or realise an appropriate value for any material future transactions. Moreover, the limited liquidity in the Ordinary Shares makes it challenging for Shareholders of any size to acquire additional Ordinary Shares or dispose of any Ordinary Shares in the market at an attractive price.

- Access to appropriate finance: The nature of the Company's operations requires the Company to periodically raise funding for working capital as the Company develops its asset portfolio. The Board considers that the future value of the Company's portfolio of assets will continue to grow as the Company invests further in its development and that significant external funding is required to ensure this development is achieved. The Board has concluded that as a private limited company it will have broader access to specialty investors and enhance the ability of the Company to raise the capital required to increase the value of its product portfolio for the benefit of all Shareholders.
- **Corporate and strategic flexibility**: The Board believes that a private limited company can take and implement strategic decisions more quickly than a company which is publicly traded as a result of the more flexible regulatory regime that is applicable to a private company. This will be advantageous in the Company's business development discussions which may ultimately benefit the Company and Shareholders as a whole.
- **Costs and regulatory burden**: The considerable cost and management time and the legal and regulatory burden associated with maintaining the Company's admission to trading on AIM is, in the Board's opinion, disproportionate to the benefits of the Company's continued admission to trading on AIM, particularly given the limited and inconsistent liquidity in the Ordinary Shares as described above. Given the lower costs associated with private limited company status, the Cancellation and Re-registration will reduce the Company's recurring administrative and adviser costs which the Board believes can be better spent supporting and investing in the Group's business.

# Therefore, as a result of this review, the Board has unanimously concluded that the proposed Cancellation and Re-registration are in the best interests of the Group and its Shareholders as a whole.

The Company remains committed to the UK life sciences industry and intends to retain its head office and primary research facility following the Cancellation and Re-registration at its current location in Alderley Park. Following the Cancellation and Re-registration, the Company will continue to evaluate the optimal corporate structure to ensure its long-term success, which could include listing on an alternative exchange at a future date, should this provide appropriate access to capital and liquidity to support the Company's strategy.

# 3. **Process for, and principal effects of, the Cancellation**

The Directors are aware that certain Shareholders may be unable or unwilling to hold Ordinary Shares in the event that the Cancellation is approved and becomes effective. Such Shareholders should consider selling their interests in the market prior to the Cancellation becoming effective. However, should the Cancellation become effective, the Company will implement a Matched Bargain Facility with a third party which would facilitate Shareholders buying and selling Ordinary Shares on a matched bargain basis following Cancellation.

Under the AIM Rules, the Company is required to give at least 20 clear Business Days' notice of the Cancellation. Additionally, the Cancellation will not take effect until at least five clear Business Days have passed following the passing of the Cancellation Resolution. If the Cancellation Resolution is passed at the General Meeting, it is proposed that the last day of trading in the Ordinary Shares on AIM will be Tuesday 30 April 2024 and that the Cancellation will take effect at 7:00 a.m. on Wednesday 1 May 2024.

If the Cancellation becomes effective, Spark will cease to be the nominated adviser of the Company and the Company will no longer be required to comply with the AIM Rules.

Under the AIM Rules, it is a requirement that the Cancellation must be approved by Shareholders holding not less than 75 per cent. of votes cast by Shareholders at the General Meeting. Accordingly, the Notice of General Meeting set out in at the end of this Document contains a special resolution to approve the Cancellation.

The principal effects of the Cancellation will include the following:

- there will be no formal market mechanism enabling Shareholders to trade Ordinary Shares (other than any limited off-market mechanism provided by the Matched Bargain Facility) and no price will be publicly quoted for the Ordinary Shares;
- it is possible that, following the publication of this Document, the liquidity and marketability of the Ordinary Shares may be significantly reduced and their value adversely affected (however, as set out above, the Directors believe that the existing liquidity in the Ordinary Shares is, in any event, limited);
- the Ordinary Shares may be more difficult to sell compared to shares of companies traded on AIM (or any other recognised market or trading exchange);
- in the absence of a formal market and quoted price, it may be difficult for Shareholders to determine the market value of their investment in the Company at any given time;
- the regulatory and financial reporting regime applicable to companies whose shares are admitted to trading on AIM will no longer apply;
- Shareholders will no longer be afforded the protections given by the AIM Rules, such as the requirement to be notified of price sensitive information or certain events and the requirement that the Company seek shareholder approval for certain corporate actions, where applicable, including substantial transactions, reverse takeovers, related party transactions and fundamental changes in the Company's business, including certain acquisitions and disposals;
- the levels of disclosure and corporate governance within the Company may not be as stringent as for a company quoted on AIM;
- the Company will no longer be subject to UK MAR regulating inside information and other matters;
- the Company will no longer be required to publicly disclose any change in major shareholdings in the Company under the Disclosure Guidance and Transparency Rules;
- the Takeover Code is expected to cease to apply to the Company shortly following the Cancellation and the Re-registration given anticipated Board changes;
- Spark will cease to be nominated adviser to the Company;
- whilst the Company's CREST facility will remain in place immediately post the Cancellation, the Company's CREST facility may be cancelled in the future and, although the Ordinary Shares will remain transferable, they may cease to be transferable through CREST (in which case, Shareholders who hold Ordinary Shares in CREST will receive share certificates);
- stamp duty will be due on transfers of shares and agreements to transfer shares unless a relevant exemption or relief applies to a particular transfer; and
- the Cancellation and Re-registration may have personal taxation consequences for Shareholders. Shareholders who are in any doubt about their tax position should consult their own professional independent tax adviser.

# The above considerations are not exhaustive, and Shareholders should seek their own independent advice when assessing the likely impact of the Cancellation on them.

For the avoidance of doubt, the Company will remain registered with the Registrar of Companies in England & Wales in accordance with, and subject to, the Companies Act, notwithstanding the Cancellation and Re-registration.

The Resolutions to be proposed at the General Meeting include the adoption of the New Articles, with effect from the Re-registration. A copy of the New Articles can be viewed at www.redxpharma.com and is included at Appendix 1 to this Document.

# 4. Board composition and provision of information, services and facilities following the Cancellation

# 4.1. Board composition

Although such changes have not yet been finally determined, the composition of the Board is expected to change shortly following the Cancellation and Re-registration in a manner appropriate for a private limited company.

As described in this Document (and, in particular, in paragraph 7 of this letter and Part III of this Document), the Takeover Code will continue to apply to the Company for a period of at least ten years from the date of the Cancellation if the Company is considered by the Panel to have its place of central management and control in the United Kingdom, the Channel Islands or the Isle of Man. This is known as the "residency test". In determining whether the residency test is satisfied, the Panel has regard primarily to whether a majority of a company's directors are resident in these jurisdictions. As noted above, the composition of the Board is expected to change shortly following the Cancellation and Re-registration and it is expected that such changes will mean that the Company no longer meets the "residency test" from that time.

#### 4.2. **Provision of information, services and facilities following the Cancellation**

The Company currently intends to continue to provide certain information, services and facilities to Shareholders following the Cancellation. The Company will:

- continue to communicate information about the Company (including annual accounts) to its Shareholders, as required by the Companies Act;
- continue, for at least 12 months following the Cancellation, to maintain its website, www.redxpharma.com and to post updates on the website from time to time, although Shareholders should be aware that there will be no obligation on the Company to include all of the information required under the Disclosure Guidance and Transparency Rules, AIM Rule 26 or to update the website as currently required by the AIM Rules; and
- make available to Shareholders, through J P Jenkins, the Matched Bargain Facility (as further described in paragraph 5.2 of this letter) which will allow Shareholders to buy and sell Ordinary Shares on a matched bargain basis following the Cancellation.

#### 5. Transactions in the Ordinary Shares prior to and post the proposed Cancellation

#### 5.1. *Prior to the Cancellation*

Shareholders should note that they are able to continue trading in the Ordinary Shares on AIM prior to the Cancellation.

#### 5.2. Following the Cancellation

The Company is making arrangements for a Matched Bargain Facility to assist Shareholders to trade in the Ordinary Shares to be put in place from the date of the Cancellation, if the Resolutions are passed. The Matched Bargain Facility will be provided by J P Jenkins. J P Jenkins is an appointed representative of Prosper Capital LLP, which is authorised and regulated by the FCA.

Under the Matched Bargain Facility, Shareholders or persons wishing to acquire or dispose of Ordinary Shares will be able to leave an indication with J P Jenkins, through their stockbroker (J P Jenkins is unable to deal directly with members of the public), of the number of Ordinary Shares that they are prepared to buy or sell at an agreed price. In the event that J P Jenkins is able to match that order with an opposite sell or buy instruction, it would contact both parties and then effect the bargain (trade). Shareholdings remain in CREST and can be traded during normal business hours via a UK regulated stockbroker. Should the Cancellation become effective and the Company puts in place the Matched Bargain Facility, details will be made available to Shareholders on the Company's website at www.redxpharma.com.

The Matched Bargain Facility will operate for a minimum of 12 months after the Cancellation. The Directors' current intention is that it will continue beyond that time but Shareholders should note that it could be withdrawn and therefore inhibit the ability to trade the Ordinary Shares. Further details will be communicated to the Shareholders at the relevant time.

# If Shareholders wish to buy or sell Ordinary Shares on AIM they must do so prior to the Cancellation becoming effective. As noted above, in the event that Shareholders approve the Cancellation, it is anticipated that the last day of dealings in the Ordinary

# Shares on AIM will be Tuesday 30 April 2024 and that the effective date of the Cancellation will be Wednesday 1 May 2024.

#### 6. **Process for the Re-Registration**

As set out above, following the Cancellation, the Directors believe that the requirements and associated costs of the Company maintaining its public company status will be difficult to justify and that the Company will benefit from the more flexible requirements and lower costs associated with private limited company status. It is therefore proposed to re-register the Company as a private limited company. In connection with the Re-registration, it is proposed that the New Articles be adopted to reflect the change in the Company's status to a private limited company. The principal effects of the Re-registration and the adoption of the New Articles on the rights and obligations of Shareholders and the Company are summarised in Part II of this Document. A copy of the New Articles can be found at Appendix 1 to this Document.

Under the Companies Act, the Re-registration and the adoption of the New Articles must be approved by Shareholders holding not less than 75 per cent. of votes cast by Shareholders at the General Meeting. Accordingly, the Notice of General Meeting set out at the end of this Document contains a special resolution to approve the Re-registration and adopt the New Articles.

If the Cancellation Resolution and the Re-registration Resolution are approved at the General Meeting, an application will be made to the Registrar of Companies for the Company to be re-registered as a private limited company. Re-registration will take effect when the Registrar of Companies issues a certificate of incorporation on Re-registration. The Registrar of Companies will issue the certificate of incorporation on Re-registration when it is satisfied that no valid application can be made to cancel the Re-registration Resolution or that any such application to cancel the Re-registration Resolution has been determined and confirmed by the Court.

If the Resolutions are passed at the General Meeting, it is anticipated that the Re-registration will become effective before the end of May 2024.

#### 7. Takeover Code

The Takeover Code applies to all offers for companies which have their registered offices in the United Kingdom, the Channel Islands or the Isle of Man if any of their equity share capital or other transferable securities carrying voting rights are admitted to trading on a UK regulated market or a UK multilateral trading facility or on any stock exchange in the Channel Islands or the Isle of Man.

The Takeover Code also applies to all offers for companies (both public and private) which have their registered offices in the United Kingdom, the Channel Islands or the Isle of Man and which are considered by the Panel to have their place of central management and control in the United Kingdom, the Channel Islands or the Isle of Man, but in relation to private companies only if one of a number of conditions is met – for example, if the company's shares were admitted to trading on a UK regulated market or a UK multilateral trading facility or on any stock exchange in the Channel Islands or the Isle of Man at any time in the preceding ten years.

If the Cancellation and Re-registration are approved by Shareholders at the General Meeting, the Company will be re-registered as a private company and its securities will no longer be admitted to trading on a regulated market or a multilateral trading facility in the United Kingdom. In these circumstances, the Takeover Code will only apply to the Company if it is considered by the Panel to have its place of central management and control in the United Kingdom, the Channel Islands or the Isle of Man. This is known as the "residency test". In determining whether the residency test is satisfied, the Panel has regard primarily to whether a majority of a company's directors are resident in these jurisdictions.

The Panel has confirmed to the Company that, on the basis of the current residency of the Directors, the Company will have its place of central management and control in the United Kingdom following the Cancellation and Re-registration. However, the composition of the Board is expected to change shortly following the Cancellation and Re-registration and it is expected that such changes will mean that the Company would no longer have its place of central management and control in the United Kingdom for the purposes of the Takeover Code from that time. Therefore, if the Cancellation and Re-registration are approved by Shareholders at the General Meeting and become effective, the Takeover Code will cease to apply to the Company with effect from such changes to the composition of the Board being made.

The Takeover Code could apply to the Company in the ten-year period from the date of the Cancellation if the composition of the Board were to change again such that the Company would have its place of central management and control in the United Kingdom. Following the expiry of the tenyear period from the date of the Cancellation (subject to Re-registration occurring), the Company would not in any circumstances be subject to the provisions of the Takeover Code.

Following either (i) any change, as expected, to the composition of the Board following the Cancellation and Re-registration such that the Company would no longer have its place of central management and control in the United Kingdom or (ii) the expiry of the ten-year period from the date of the Cancellation (subject to Re-registration occurring), the Code will cease to apply to the Company and Shareholders will no longer be afforded the protections provided by the Code. This includes the requirement for a mandatory cash offer to be made if either:

- a person acquires an interest in shares which, when taken together with the shares in which persons acting in concert with it are interested, increases the percentage of shares carrying voting rights in which it is interested to 30 per cent. or more; or
- a person, together with persons acting in concert with it, is interested in shares which in the aggregate carry not less than 30 per cent. of the voting rights of a company but does not hold shares carrying more than 50 per cent. of such voting rights and such person, or any person acting in concert with it, acquires an interest in any other shares which increases the percentage of shares carrying voting rights in which it is interested.

Brief details of the Panel, and of the protections afforded by the Takeover Code are set out in Part III of this Document.

# Before giving your consent to the Cancellation and the Re-registration, you may want to take independent professional advice from an appropriate independent financial adviser.

#### 8. General Meeting

The General Meeting will be held at the offices of Cooley (UK) LLP, 22 Bishopsgate, London EC2N 4BQ at 11:30 a.m. on Friday 19 April 2024.

Resolution 1 to be proposed at the General Meeting is a special resolution to approve the Cancellation.

Conditional on the passing of Resolution 1, Resolution 2 to be proposed at the General Meeting is a special resolution to re-register the Company as a private limited company and to approve the adoption by the Company of the New Articles.

Resolution 1 to approve the Cancellation is not conditional on Resolution 2 to approve the Reregistration, but Resolution 2 is conditional on Resolution 1. If Resolution 1 is passed, but Resolution 2 is not, the Company still intends to proceed with the Cancellation.

# 9. Irrevocable Undertakings and Letters of Intent

Those Directors who hold Ordinary Shares, who in aggregate hold 394,154 Ordinary Shares, representing approximately 0.10 per cent. of the Existing Ordinary Shares, have irrevocably undertaken to vote in favour of the Resolutions at the General Meeting.

In addition, the Company has received letters of intent from RedCo II, RM3 and Sofinnova who in aggregate hold 328,838,626 Ordinary Shares representing approximately 84.54 per cent. of the Existing Ordinary Shares, confirming that each of such Shareholders intends to cast, or procure that all the votes attaching to the Ordinary Shares held by such Shareholders are cast, in favour of the Resolutions at the General Meeting.

# 10. Action to be taken in relation to the General Meeting

A Form of Proxy for use at the General Meeting is enclosed with this Document. The Form of Proxy should be completed in accordance with the instructions printed thereon and returned to Equiniti Limited, Aspect House, Spencer Road, Lancing, West Sussex BN99 6DA, U.K. as soon as possible but in any event by no later than 11:30 a.m. on Wednesday 17 April 2024. Shareholders who hold their Ordinary Shares in uncertificated form in CREST may alternatively use the CREST proxy voting service in accordance with the procedures set out in the CREST Manual as explained in the notes

accompanying the Notice of General Meeting at the end of this document. Proxies submitted via CREST must be received by the Company's registrars, Equiniti Limited, by no later than 11:30 a.m. on Wednesday 17 April 2024.

The release, publication or distribution of this Document and the Form of Proxy in jurisdictions other than the UK may be restricted by laws or regulations and therefore persons into whose possession this Document and/or the Form of Proxy come, should inform themselves about, and observe, any such restrictions. Any failure to comply with any such restrictions may constitute a violation of the securities laws or regulations of such jurisdictions.

Before deciding what action to take in respect of the Resolutions, you are advised to read the whole of this Document and not merely rely on certain sections of this Document. If you are in any doubt as to the action you should take, you should immediately seek your own personal financial advice from an appropriately qualified independent professional adviser.

Shareholders are encouraged to appoint the chair of the General Meeting as their proxy with directions as to how to cast their vote on the Resolutions proposed. The appointment of a proxy will not preclude Shareholders from attending and voting at the General Meeting in person should they so wish.

It is important that as many votes as possible are cast. Whether or not you plan to attend the General Meeting in person, you are encouraged to complete and return your Form of Proxy as soon as possible.

#### 11. Recommendation

The Directors consider that the Resolutions are in the best interests of the Company and its Shareholders as a whole. Accordingly, the Directors unanimously recommend that you vote in favour of the Resolutions as they intend to do in respect of their own shareholdings of 394,154 Ordinary Shares, representing approximately 0.10 per cent. of the Existing Ordinary Shares.

Yours faithfully,

**Dr Jane Griffiths** *Chair of the Board of Directors* 

# PART II

# PRINCIPAL EFFECT OF RE-REGISTRATION AND ADOPTION OF NEW ARTICLES ON SHAREHOLDERS

# 1. Accounts

A public company is required to file its accounts within six months following the end of its financial year and then to circulate copies of the accounts to shareholders. Following the Re-registration and the adoption of the New Articles, the period for the preparation of accounts is extended to nine months following the end of the financial year. The Company will still be required to circulate accounts to Shareholders (although the period for doing so is extended for private companies).

# 2. General meetings and resolutions

A public company is required to hold an annual general meeting of shareholders each year, whereas a private company is not. Therefore, following the Re-registration and the adoption of the New Articles the Company will not hold annual general meetings.

In addition, after the Re-registration, resolutions of the Shareholders of the Company may be obtained via written resolutions, rather than via physical meetings. This is done by obtaining the approval in writing to that resolution of the holders of a majority of voting shares then in issue (in the case of ordinary resolutions) and the holders of at least 75 per cent. of the voting shares then in issue (in the case of special resolutions).

# 3. Directors

The Current Articles contain provisions requiring: (a) at the annual general meeting of the Company, one-third of the Directors (or, if their number is not three or a multiple of three, then the nearest number to but not less than one-third) retire from office and (b) that each Director retire at least once every three years. These provisions are not included in the New Articles. In addition, the New Articles will not require any director appointed by the Board to be re-appointed by the Shareholders at the next annual general meeting following his or her appointment, as is currently required.

# 4. Issue of shares for non-cash consideration

As a public company, there are restrictions on the ability of the Company to issue new shares, for example, by requiring the Company to obtain a valuation report in the case of shares issued for non-cash consideration. These restrictions will not apply following the Re-registration and adoption of the New Articles.

# 5. Financial assistance, reductions of capital and purchase of own shares out of capital

As a public limited company, the Company is currently prohibited from performing actions which constitute financial assistance for the acquisition of its own shares. This limits the ability of the Company to engage in certain transactions. However, following the Re-registration, these restrictions will no longer apply.

In addition, the Company must currently obtain the sanction of the Court for any reduction of capital, which can be a lengthy and expensive process. However, following the Re-registration, the Company will be able to take advantage of more flexible provisions applicable to private companies, which do not require the approval of the Court.

Similarly, following the Re-registration, the Company will be able to effect purchases of its own shares out of capital, which it is currently prohibited from doing as a public limited company.

# 6. Company secretary

There is no requirement for a company secretary to be appointed, although the Company may appoint one should it wish.

# 7. Drag along and tag along provisions

The New Articles contain drag along and tag along provisions which would apply in the case of certain proposed transfers of shares.

# 8. Removal of unnecessary provisions and simplification

The New Articles will not contain certain of the detailed provisions of the Current Articles which are common for listed companies, and which will not be necessary for the Company following the Cancellation. The New Articles are in a form appropriate for a private limited company.

# PART III

# THE TAKEOVER CODE

The Takeover Code currently applies to the Company and will do so for ten years following the Cancellation becoming effective and the Re-registration occurring if the Company meets the "residency test" under the Takeover Code. The composition of the Board is expected to change shortly following the Cancellation and Re-registration such that the Company would no longer meet the residency test from that time. Following the Cancellation becoming effective and the Re-registration occurring, once the Company no longer meets the residency test the Takeover Code will not apply to the Company and the protections afforded by the Takeover Code will not apply to any offer made to Shareholders to acquire their Ordinary Shares.

The Takeover Code applies to all offers for companies which have their registered offices in the United Kingdom, the Channel Islands or the Isle of Man if any of their equity share capital or other transferable securities carrying voting rights are admitted to trading on a UK regulated market or a UK multilateral trading facility or on any stock exchange in the Channel Islands or the Isle of Man.

The Takeover Code also applies to all offers for companies (both public and private) which have their registered office in the United Kingdom, the Channel Islands or the Isle of Man which are considered by the Panel to have their place of central management and control in the United Kingdom, the Channel Islands or the Isle of Man, but in relation to private companies only if one of a number of conditions is met – for example, if the company's shares were admitted to trading on a UK regulated market or a UK multilateral trading facility or on any stock exchange in the Channel Islands or the Isle of Man at any time in the preceding ten years.

If the Cancellation and Re-registration are approved by Shareholders at the General Meeting, the Company will be re-registered as a private company and its securities will no longer be admitted to trading on a regulated market or a multilateral trading facility in the United Kingdom. In these circumstances, the Takeover Code will only apply to the Company if it is considered by the Panel to have its place of central management and control in the United Kingdom, the Channel Islands or the Isle of Man. This is known as the "residency test". In determining whether the residency test is satisfied, the Panel has regard primarily to whether a majority of a company's directors are resident in these jurisdictions.

The Panel has confirmed to the Company that, on the basis of the current residency of the Directors, the Company will have its place of central management and control in the United Kingdom following the Cancellation and Re-registration. However, the composition of the Board is expected to change shortly following the Cancellation and Re-registration and it is expected that such changes will mean that the Company would no longer have its place of central management and control in the United Kingdom for the purposes of the Takeover Code from that time. Therefore, if the Cancellation and Re-registration are approved by Shareholders at the General Meeting and become effective, the Takeover Code will cease to apply to the Company with effect from such changes to the composition of the Board being made.

The Takeover Code could apply to the Company in the ten-year period from the date of the Cancellation if the composition of the Board were to change again such that the Company would have its place of central management and control in the United Kingdom. Following the expiry of the tenyear period from the date of the Cancellation (subject to the Re-registration occurring), the Company would not in any circumstances be subject to the provisions of the Takeover Code.

Brief details of the Panel, and of the protections afforded by the Takeover Code are described below.

Before giving your consent to the Cancellation and the Re-registration, you may want to take independent professional advice from an appropriate independent financial adviser.

#### The Takeover Code

The Takeover Code is issued and administered by the Panel. The Takeover Code currently applies to the Company and, accordingly, its Shareholders are entitled to the protections afforded by the Takeover Code.

The Takeover Code and the Panel operate principally to ensure that shareholders are treated fairly and are not denied an opportunity to decide on the merits of a takeover and that shareholders of the same class are afforded equivalent treatment by an offeror. The Takeover Code also provides an orderly framework within which takeovers are conducted. In addition, it is designed to promote, in conjunction with other regulatory regimes, the integrity of the financial markets.

#### The General Principles and Rules of the Takeover Code

The Takeover Code is based upon a number of General Principles which are essentially statements of standards of commercial behaviour. For your information, these General Principles are set out in Part 1 of Appendix A of this Part III. The General Principles apply to all transactions with which the Takeover Code is concerned. They are applied by the Panel in accordance with their spirit to achieve their underlying purpose.

In addition to the General Principles, the Takeover Code contains a series of Rules. Some of the Rules provide more detail on how the General Principles will be applied by the Panel and others govern specific aspects of the takeover procedure. Like the General Principles, the Rules are to be interpreted to achieve their underlying purpose. Therefore, their spirit must be observed as well as their letter. The Panel may derogate or grant a waiver to a person from the application of a Rule in certain circumstances.

#### Giving up the protection of the Takeover Code

A summary of key points regarding the application of the Takeover Code to takeovers generally is set out in Part 2 of Appendix A of this Part III. You are encouraged to read this information carefully as it outlines certain important protections which you will be giving up if you agree to the Cancellation and the Re-registration and, as expected, the Takeover Code ceases to apply to the Company in the future.

# **APPENDIX A**

# PART 1: THE GENERAL PRINCIPLES OF THE TAKEOVER CODE

- 1. All holders of the securities of an offeree company of the same class must be afforded equivalent treatment. If a person acquires control of a company, the other holders of securities must be protected.
- 2. The holders of the securities of an offeree company must have sufficient time and information to enable them to reach a properly informed decision on the takeover bid. Where it advises the holders of securities, the board of directors of the offeree company must give its views on the effects of implementation of the takeover bid on employment, conditions of employment and the locations of the company's places of business.
- 3. The board of directors of an offeree company must act in the interests of the company as a whole and must not deny the holders of securities the opportunity to decide on the merits of the takeover bid.
- 4. False markets must not be created in the securities of the offeree company, of the offeror company, or of any other company concerned by the takeover bid in such a way that the rise or fall of the prices of the securities becomes artificial and the normal functioning of the markets is distorted.
- 5. An offeror must announce a takeover bid only after ensuring that the offeror can fulfil in full any cash consideration, if such is offered, and after taking all reasonable measures to secure the implementation of any other type of consideration.
- 6. An offeree company must not be hindered in the conduct of its affairs for longer than is reasonable by a takeover bid for its securities.

# PART 2: DETAILED APPLICATION OF THE TAKEOVER CODE

The following is a summary of key provisions of the Takeover Code which apply to transactions to which the Takeover Code applies. You should note that, by agreeing to the Cancellation and the Re-registration, you will be giving up protections afforded by the Takeover Code in the event that, as expected, the Takeover Code ceases to apply to the Company in the future.

#### Equality of treatment

General Principle 1 of the Takeover Code states that all holders of securities of an offeree company of the same class must be afforded equivalent treatment. Furthermore, Rule 16.1 requires that, except with the consent of the Panel, special arrangements may not be made with certain shareholders in the Company if there are favourable conditions attached which are not being extended to all shareholders.

#### Information to shareholders

General Principle 2 requires that holders of securities of an offeree company must have sufficient time and information to enable them to reach a properly informed decision on a takeover bid. Consequently, a document setting out full details of an offer must be sent to the offeree company's shareholders.

#### The opinion of the offeree board and independent advice

The board of the offeree company is required by Rule 3.1 of the Takeover Code to obtain competent independent advice as to whether the financial terms of an offer are fair and reasonable and the substance of such advice must be made known to its shareholders. Rule 25.2 requires that the board of the offeree company must send to the offeree company's shareholders and persons with information rights its opinion on the offer and its reasons for forming that opinion. That opinion must include the board's views on: (i) the effects of implementation of the offer on all the company's interests, including, specifically, employment; and (ii) the offeror's strategic plans for the offeree company and their likely repercussions on employment and the locations of the offeree company's places of business.

The document sent to shareholders must also deal with other matters such as interests and recent dealings in the securities of the offeror and the offeree company by relevant parties and whether the

directors of the offeree company intend to accept or reject the offer in respect of their own beneficial shareholdings.

Rule 20.1 states that, except in certain circumstances, information and opinions relating to an offer or a party to an offer must be made equally available to all offeree company shareholders and persons with information rights as nearly as possible at the same time and in the same manner.

#### Option-holders and holders of convertible securities or subscription rights

Rule 15 of the Takeover Code provides that when an offer is made and the offeree company has convertible securities outstanding, the offeror must make an appropriate offer or proposal to the holders of those securities to ensure that their interests are safeguarded. Rule 15 also applies in relation to holders of options and other subscription rights. If Cancellation and Re-registration occurs, these protections will be lost in the event that the Takeover Code ceases to apply to the Company in the future.

# NOTICE OF GENERAL MEETING

# **REDX PHARMA PLC**

(incorporated and registered in England and Wales with registered number 7368089) (the "**Company**")

**NOTICE IS HEREBY GIVEN THAT** a general meeting of the Company (the "**General Meeting**") will be held at 11:30 a.m. on Friday 19 April 2024 at the offices of Cooley (UK) LLP, 22 Bishopsgate, London EC2N 4BQ to consider and, if thought fit, approve the special resolutions set out below.

In this notice (the "**Notice**"), unless otherwise defined, words and defined terms shall have the same meaning as words and defined terms in the Document to which this Notice is attached.

# SPECIAL RESOLUTIONS

- 1. **THAT**, in accordance with Rule 41 of the AIM Rules for Companies, the cancellation of the admission to trading on AIM (the market of that name operated by London Stock Exchange plc) of the ordinary shares of £0.01 each in the capital of the Company be and is hereby approved and the directors of the Company be and are hereby authorised to take all action reasonable or necessary to effect such cancellation.
- 2. **THAT**, subject to and conditional upon Resolution 1 proposed at the General Meeting being approved at the General Meeting and the cancellation of the admission of the ordinary shares of £0.01 in the capital of the Company to trading on AIM (the market of that name operated by London Stock Exchange plc) becoming effective:
  - (a) the Company be re-registered as a private limited company under the Companies Act 2006 with the name of Redx Pharma Limited; and
  - (b) the regulations contained in the document submitted to the General Meeting and for the purposes of identification initialled by or on behalf of the Chair of the General Meeting be approved and adopted as the articles of association of the Company in substitution for and to the exclusion of the existing articles of association with effect from the issue of the certificate of incorporation as a private limited company.

By order of the Board Claire Solk Company Secretary Registered Office: Block 33, Mereside Alderley Park Macclesfield SK10 4TG

2 April 2024

#### **EXPLANATORY NOTES TO THE NOTICE OF GENERAL MEETING:**

- 1. The following notes explain your general rights as a Shareholder and your rights to attend and vote at the General Meeting or to appoint someone else to vote at the General Meeting on your behalf.
- 2. A Shareholder entitled to attend and vote at the General Meeting is also entitled to appoint one or more proxies to attend, speak and vote on a show of hands and on a poll instead of him or her. A proxy need not be a Shareholder. Where a Shareholder appoints more than one proxy, each proxy must be appointed in respect of different Ordinary Shares comprised in his or her shareholding which must be identified on the Form of Proxy. Each such proxy will have the right to vote on a poll in respect of the number of votes attaching to the number of Ordinary Shares in respect of which the proxy has been appointed. Where more than one joint Shareholder purports to appoint a proxy in respect of the same Ordinary Shares, only the appointment by the most senior Shareholder will be accepted as determined by the order in which their names appear in the Company's register of members. If you wish your proxy to speak at the General Meeting, you should appoint a proxy other than the Chair of the General Meeting and give your instructions to that proxy.
- 3. A Shareholder may appoint more than one proxy in relation to the General Meeting, provided that each proxy is appointed to exercise the rights attached to different Ordinary Shares held by him or her. A Form of Proxy is enclosed with this Notice. Shareholders who intend to appoint more than one proxy may photocopy the Form of Proxy prior to completion. Alternatively, additional Forms of Proxy may be obtained by contacting Equiniti Limited on +44 (0) 371 384 2030. If calling from outside the United Kingdom, please ensure the country code is used. The Forms of Proxy should be returned in the same envelope and each should indicate that it is one or more than one appointments being made. To be valid, a Form of Proxy together with, if applicable, the power of attorney or other authority under which it is signed, or a certified copy thereof, must be received by Equiniti Limited, Aspect House, Spencer Road, Lancing, West Sussex BN99 6DA not later than 11:30 a.m. on 17 April 2024 (or not less than 48 hours before the time of any adjourned meeting).
- 4. An abstention (or "vote withheld") option has been included on the Form of Proxy. The legal effect of choosing the abstention option on any resolution is that the Shareholder concerned will be treated as not having voted on the relevant resolution. The number of votes in respect of which there are abstentions will however be counted and recorded, but disregarded in calculating the number of votes for or against each resolution.
- 5. Completion and return of the Form of Proxy or any CREST Proxy Instruction (as defined below) will not preclude a Shareholder from attending and voting in person at the General Meeting or any adjournment thereof should he/she wish to do so.
- 6. Any person to whom this Notice is sent who is a person nominated under section 146 of the Companies Act to enjoy information rights (a "Nominated Person") may, under an agreement between him/her and the Shareholder by whom he/she was nominated, have a right to be appointed (or to have someone else appointed) as a proxy for the General Meeting. If a Nominated Person has no such proxy appointment right or does not wish to exercise it, he/she may, under any such agreement, have a right to give instructions to the Shareholder as to the exercise of voting rights.
- 7. The statement of rights of Shareholders in relation to the appointment of proxies in paragraphs 2 and 3 above does not apply to Nominated Persons. The rights described in these paragraphs can only be exercised by Shareholders.
- 8. Any corporation which is a Shareholder can appoint one or more corporate representatives. Each representative may exercise on behalf of the corporation the same powers as the corporation could exercise if it were an individual Shareholder provided that they do not do so in relation to the same Ordinary Share. It is therefore no longer necessary to nominate a designated corporate representative.
- 9. The Company, pursuant to Regulation 41 of the CREST Regulations, specifies that only those Shareholders registered in the register of members of the Company as at 6:30 p.m. on 17 April 2024 shall be entitled to attend or vote (whether on a show of hands or on a poll) at the General Meeting in respect of the number of Ordinary Shares registered in their name at that time. Changes to entries on the register after 6:30 p.m. on 17 April 2024 (or after 6:30 p.m. on the day which is two days before any adjourned meeting, excluding non-working days) shall be disregarded in determining the rights of any person to attend or vote at the meeting.
- 10. CREST members who wish to appoint a proxy or proxies through the CREST electronic proxy appointment service may do so for the General Meeting to be held at 11:30 a.m. on 19 April 2024 at the offices of Cooley (UK) LLP, 22 Bishopsgate, London EC2N 4BQ and any adjournment(s) thereof by using the procedures described in the CREST Manual available via www.euroclear.com. CREST personal members or other CREST sponsored members, and those CREST members who have appointed a voting service provider(s), should refer to their CREST sponsors or voting service provider(s), who will be able to take the appropriate action on their behalf.

- 11. In order for a proxy appointment or instruction made by means of CREST to be valid, the appropriate CREST message (a "**CREST Proxy Instruction**") must be properly authenticated in accordance with Euroclear's specifications and must contain the information required for such instructions, as described in the CREST Manual available via www.euroclear.com. The message, regardless of whether it constitutes the appointment of a proxy or an amendment to the instruction given to a previously appointed proxy, must be transmitted so as to be received by the Company's agent, Equiniti Limited (CREST Participant ID: RA19), no later than 48 hours before the time appointed for the General Meeting (excluding non-working days). For this purpose, the time of receipt will be taken to be the time (as determined by the time stamp applied to the message by the CREST Application Host) from which the Company's agent is able to retrieve the message by enquiry to CREST in the manner prescribed by CREST. After this time any change of instructions to proxies appointed through CREST should be communicated to the appointee through other means.
- 12. CREST members and, where applicable, their CREST sponsor or voting service provider should note that Euroclear does not make available special procedures in CREST for any particular messages. Normal system timings and limitations will therefore apply in relation to the input of CREST Proxy Instructions. It is the responsibility of the CREST member concerned to take (or, if the CREST member is a CREST personal member or sponsored member or has appointed a voting service provider(s), to procure that his CREST sponsor or voting service provider(s) take(s)) such action as shall be necessary to ensure that a message is transmitted by means of the CREST system by any particular time. In this connection, CREST members and, where applicable, their CREST sponsor or voting service provider are referred in particular to those sections of the CREST Manual concerning practical limitations of the CREST system and timings.
- 13. The Company may treat as invalid a CREST Proxy Instruction in the circumstances set out in Regulation 35(5)(a) of the CREST Regulations.
- 14. Shareholders, proxies and authorised representatives will be required to provide their names and addresses for verification against the register of members and proxy appointments received by the Company before entering the General Meeting. Each authorised representative must produce proof of his or her appointment, in the form of the actual appointment or a certified copy. Other than this, there are no procedures with which any such persons must comply in order to attend and vote at the General Meeting.
- 15. Shareholders, proxies and authorised representatives may raise questions at the General Meeting concerning the business being dealt with at the General Meeting and will receive answers, except that a question need not be answered where it would interfere unduly with the conduct of the General Meeting, would involve the disclosure of confidential information, where the answer has already been given on a website in the form of an answer to a question or where it is undesirable in the interests of the Company or the good order of the General Meeting that the question be answered.
- 16. As at 28 March 2024 (being the latest practicable date prior to the date of this Document), the Company's issued share capital consisted of 388,985,916 ordinary shares of £0.01 each, each carrying the right to one vote at a general meeting of the Company. As at the date of this Document, the Company does not hold any ordinary shares in treasury. Therefore, the total number of voting rights in the Company as at 28 March 2024 was 388,985,916.
- 17. A copy of this Notice is available on the Company's website at https://www.redxpharma.com/investors/investor-centre/.

# **APPENDIX 1: NEW ARTICLES**

Company Number: 07368089

# PRIVATE COMPANY LIMITED BY SHARES

# **ARTICLES OF ASSOCIATION**

of

### **REDX PHARMA LIMITED**

Incorporated in England and Wales on 7 September 2010 under the Companies Act 2006

Adopted under the Companies Act 2006 by special resolution on [•] April 2024 and effective on [•] May 2024



# TABLE OF CONTENTS

# Page

1.	PRELIMINARY	1
2.	LIABILITY OF MEMBERS	4
3.	DIRECTORS' POWERS, RESPONSIBILITIES AND DELEGATION	4
4.	DECISION-MAKING BY DIRECTORS	4
5.	DIRECTORS' PERMITTED INTERESTS	8
6.	AUTHORISATION OF CONFLICTS OF INTEREST	10
7.	DIRECTORS' INTERESTS GENERAL	11
8.	ALTERNATE DIRECTORS	12
9.	APPOINTMENT, RETIREMENT AND REMOVAL OF DIRECTORS	14
10.	DIRECTORS' REMUNERATION AND EXPENSES	15
11.	SHARES: GENERAL	15
12.	UNCERTIFICATED SHARES	16
13.	SHARE CERTIFICATES	17
14.	REPLACEMENT CERTIFICATES	18
15.	SHARES: TRANSFER	18
16.	SHARES: DRAG ALONG	20
17.	SHARES: TAG ALONG	20
18.	SHARES: TRANSMISSION	21
19.	DIVIDENDS AND OTHER DISTRIBUTIONS	22
20.	CAPITALISATION OF PROFITS	24
21.	DECISION-MAKING BY SHAREHOLDERS: ORGANISATION OF GENERAL MEETINGS	24
22.	DECISION-MAKING BY SHAREHOLDERS: VOTING AT GENERAL MEETINGS	27
23.	COMPANY SECRETARY	31
24.	AUTHENTICATION	31
25.	COMPANY SEALS	31
26.	PROVISION FOR EMPLOYEES ON THE CESSATION OF BUSINESS	31
27.	NOTICES AND COMMUNICATIONS	31
28.	INDEMNITIES AND FUNDING OF PROCEEDINGS	33
29.	INSURANCE	33
30.	DATA PROTECTION	34

#### **ARTICLES OF ASSOCIATION**

#### - of -

# **REDX PHARMA LIMITED**

#### ("Company")

#### 1. **PRELIMINARY**

- 1.1 The relevant model articles (within the meaning of section 20(2) Companies Act 2006 as amended, modified or re-enacted from time to time) are excluded in their entirety.
- 1.2 For so long as there is only one shareholder of the Company (ignoring for these purposes the Company as holder of any treasury shares) references in these articles to shareholders or which imply the existence of more than one shareholder shall be construed as references to the one shareholder for the time being of the Company.
- 1.3 In these articles (unless the context requires otherwise) the following words and expressions have the following meanings:

"Accepting Shareholders" has the meaning given in article 16.1;

"Appointor" has the meaning given in article 8.1;

"Associate" in relation to any person ("first person") shall mean any person ("second person") who is connected with that first person within the meaning of sections 1122 and 1123 Corporation Tax Act 2010;

"associated company" has the meaning given in article 28.1;

**"bankruptcy"** means the making of a bankruptcy order by a court in England and Wales or Northern Ireland, or any individual insolvency proceedings in a jurisdiction other than England and Wales or Northern Ireland which have an effect similar to that of bankruptcy;

"**Business Day**" means a day (excluding Saturdays, Sundays and public holidays in England and Wales) on which banks are generally open for the transaction of normal banking business in London;

"capitalised sum" has the meaning given in article 20.1(b);

"Chairperson" has the meaning given in article 4.6(a);

"chairperson of the meeting" has the meaning given in article 21.3;

"clear days" in relation to a period of notice means a period of the specified length excluding the date on which notice is given and the day for which it is given or on which it is to take effect;

**"Companies Acts"** means the Companies Acts (as defined in section 2 of the Companies Act 2006 as amended, modified or re-enacted from time to time), in so far as they apply to the Company;

"**Companies Act 2006**" means the Companies Act 2006 including any statutory modification or re-enactment of that statute for the time being in force, subject always to article 1.3;

"**Company Secretary**" means the secretary of the Company, if any, or any other person appointed to perform the duties of the secretary of the Company, including a joint, assistant or deputy secretary, if any;

"Distribution Recipient" has the meaning given in article 19.2(b);

"**document**" includes, unless otherwise specified, any document sent or supplied in electronic form;

"Drag Along Notice" has the meaning given in article 16.1;

"eligible director" means (a) in relation to a matter proposed at a directors' meeting, a director who is entitled to vote and to have that vote counted in relation to that particular matter at that meeting, or (b) in relation to a decision of the directors taken in accordance with article 4.2 a director who would have been entitled to vote and to have that vote counted, had the matter in question been proposed at a directors' meeting;

"**fully paid**" in relation to a share, means that the nominal value and any premium to be paid to the Company in respect of that share have been paid or credited as paid to the Company;

"Group Companies" means the Company and its subsidiary undertakings from time to time, and a reference to a "Group Company" shall be a reference to any one of them;

"instrument" means a document in hard copy form;

"**Operator**" means Euroclear UK and International Limited or such other person as may for the time being be approved by HM Treasury as Operator under the uncertificated securities rules;

"Other Shareholders" has the meaning given in article 16.1;

"persons entitled" has the meaning given in article 20.1(b);

"Proposed Buyer" has the meaning given in article 17.1;

"Proposed Sale" has the meaning given in article 17.1;

"Proposed Sellers" has the meaning given in article 17.1;

"Proxy Notice" has the meaning given in article 22.5(a);

"qualifying person" has the meaning given in article 21.2;

"Register" means the register of members of the Company to be maintained under the Act;

"Relevant Company" has the meaning given in article 29.2;

"**Relevant Matter**" means in relation to a director, a matter which may constitute or give rise to a breach by that director of his or hers duty under section 175 Companies Act 2006 to avoid a situation in which he or she has, or can have, a direct or indirect interest that conflicts or possibly may conflict with the interests of the Company (including a breach which would arise by virtue of his or hers appointment as a director);

"**shareholder**" means a person whose name is entered on the register of members as the holder of a share;

"share" means a share in the Company;

"Tag Along Notice" has the meaning given in article 17.1;

"Third Party" has the meaning given in article 16.1;

"Third Party Offer" has the meaning given in article 16.1;

"**Transmittee**" means a person entitled to a share by reason of the death or bankruptcy of a shareholder or otherwise by operation of law;

"**treasury share**" means any share held by the Company as a treasury share within the meaning of section 724 Companies Act 2006;

"**uncertificated securities rules**" means any provision of the Companies Acts relating to the holding, evidencing of title to, or transfer of uncertificated shares and any legislation, rules or other arrangements made under or by virtue of such provision (including the Uncertificated Securities Regulations 2001 (*SI 2001/3755*) as amended or replaced from time to time and any subordinate legislation or rules made under them for the time being in force);

"uncertificated share" means a share of a class which is at the relevant time a participating class, title to which is recorded on the Register as being held in uncertificated form and references in these articles to a share being held in uncertificated form shall be construed accordingly;

"United Kingdom" means Great Britain and Northern Ireland; and

"writing" means a method of representing or reproducing words, symbols or other information by any method or combination of methods, whether in electronic form, hard copy or in any other legible and non-transitory form and "written" shall be construed accordingly.

Words and expressions defined in the Companies Act 2006 and used in these articles (either without further definition or by expressly referring to the statutory definition of that word or expression) shall bear the same meaning as in the Companies Act 2006 as in force on the date when these articles become binding on the Company. This does not apply (a) where the word or expression used is not defined by express reference to the Companies Act 2006 and the

subject or context in which that word or expression is used is inconsistent with the statutory definition, or (b) where that word or expression is otherwise defined in these articles. In all other circumstances references in these articles to any statute or statutory provision (including without limitation the Companies Act 2006 or any provision of the Companies Act 2006) subordinate legislation, code or guideline ("**legislation**") is a reference to such legislation as the same may from time to time be amended, re-enacted, modified, extended, varied, superseded, replaced, substituted or consolidated.

# 2. LIABILITY OF MEMBERS

The liability of the members is limited to the amount if any, unpaid on the shares held by them.

#### 3. DIRECTORS' POWERS, RESPONSIBILITIES AND DELEGATION

- 3.1 Subject to these articles, the directors are responsible for the management of the Company's business, for which purpose they may exercise all the powers of the Company.
- 3.2 The shareholders may, by special resolution, direct the directors to take, or refrain from taking, specified action. No such special resolution shall invalidate anything which the directors have done before the passing of the resolution.
- 3.3 The directors may, by a decision taken in accordance with article 4.1 or 4.2, exercise the powers of the Company to change the Company's name.
- 3.4 Subject to these articles, the directors may delegate any of the powers which are conferred on them under these articles to such person or committee, by such means (including by power of attorney) to such an extent, in relation to such matters or territories, and on such terms and conditions, as they think fit. If the directors so specify, any such delegation may authorise further delegation of the directors' powers by any person to whom they are delegated. The directors may revoke any delegation in whole or part, or alter its terms and conditions.
- 3.5 Committees to which the directors delegate any of their powers must follow procedures which are based, as far as they are applicable, on those provisions of these articles which govern the taking of decisions by directors. The directors may make rules of procedure for all or any committees which prevail over rules derived from these articles if they are not consistent with them.

# 4. DECISION-MAKING BY DIRECTORS

#### 4.1 Directors to take decisions collectively

- (a) The general rule about decision making by directors is that any decision of the directors must either be a majority decision at a meeting or taken in accordance with article 4.2.
- (b) If the Company only has one director for the time being (and no provision of these articles requires it to have more than one director) the general rule does not apply and the sole director (for as long as he or she remains the sole director) shall be entitled to exercise all the powers and authorities vested in the directors by these articles (and the provisions of these articles shall be construed accordingly), and he or she may take decisions (provided that he or she constitutes an eligible director in relation to any

particular decision) without regard to the provisions of articles 4.2, 4.3, 4.4(a), 4.4(b), 4.4(e), 4.4(f), 4.5(a), 4.5(b) and 4.6 relating to directors' decision-making.

#### 4.2 Unanimous decisions

A decision of the directors is taken in accordance with this article when all eligible directors indicate to each other by any means that they share a common view on a matter. Such a decision may take the form of a resolution in writing, where each eligible director has signed one or more copies of it, or to which each eligible director has otherwise indicated his or hers agreement in writing. A decision may only be taken in accordance with this article 4.2 where the eligible directors taking the decision would have formed a quorum had the matter been proposed as a resolution at a directors' meeting.

#### 4.3 Calling a directors' meeting

- (a) Any director may call a directors' meeting by giving notice of the meeting to the directors or by authorising the Company Secretary to give such notice. The Company Secretary must call a directors' meeting if a director so requests.
- (b) Notice of any directors' meeting must indicate its proposed location (if any), its proposed date and time and, if it is anticipated that directors participating in the meeting will not be in the same place, how it is proposed that they should communicate with each other during the meeting.
- (c) Subject to these articles, notice of a meeting of the directors must be given to each director (including one who is absent for the time being from the United Kingdom) and may be given either personally or by word of mouth or in hard copy form or by electronic means, or by any other means authorised by the director concerned.
- (d) Notice of a directors' meeting need not be given to directors who are not entitled to receive notice, or who have elected not to receive notice of that meeting pursuant to article 7.1(b), or who have waived their entitlement to notice of that meeting by giving notice to that effect to the Company in advance of the meeting or not more than 7 days after the date on which the meeting is held. Where such notice of waiver is given after the meeting has been held, that does not affect the validity of the meeting, or of any business conducted at it.

#### 4.4 **Participation in directors' meetings and decision making**

(a) Subject to these articles, the directors participate in a directors' meeting when the meeting has been called and takes place in accordance with these articles and where each director can communicate orally to all of the other directors taking part any information or opinions he or she has on any particular item of the business of the meeting. In determining whether the directors are participating in a directors' meeting it is irrelevant where any director is or (subject to the first sentence of this article) how the directors communicate with each other. Such a meeting shall be deemed to take place where the largest group of those participating is assembled, or, if there is no such group, where the chairperson of the board meeting is located.

- (b) Subject to these articles, each director participating in a directors' meeting has one vote.
- (c) Subject to the Companies Act 2006 and the other provisions of these articles, a director may participate in any decision-making process (including being able to vote on, and be counted in the quorum at any meeting) where the matter under consideration or resolution to be voted on concerns a matter in which he or she has a direct or indirect interest which conflicts or may conflict with the interests of the Company provided that:
  - the director has declared the nature and extent of that interest in accordance with and to the extent required by the provisions of the Companies Act 2006 and these articles;
  - (ii) where necessary, any situation which could give rise to a conflict and which would otherwise be prohibited by section 175 of the Companies Act 2006 is authorised pursuant to article 5.1 or article 6; and
  - (iii) the terms of any authorisation given or imposed pursuant to article 5.1 or article 6 do not prevent or otherwise restrict the director from doing so,

but otherwise shall not be entitled to participate in such process or to vote or count in the quorum where he or she has a direct or indirect interest which conflicts or may conflict with the interests of the Company. If a director purports to vote in a situation where, by virtue of this article 4.4(c) (or the terms of any authorisation) he or she is not so entitled, his or hers vote shall not be counted.

- (d) For the purposes of article 4.4 (c):
  - an interest of a person who is connected with a director (within the meaning of section 252 of the Companies Act 2006) shall be treated as an interest of the director;
  - (ii) in relation to an alternate, an interest of his or hers Appointor shall be treated as an interest of the alternate in addition to any interest which the alternate otherwise has, but this does not preclude the alternate from voting in relation to that transaction or arrangement on behalf of another Appointor who does not have such an interest (or for himself if he or she is a director and has no such interest);
  - (iii) references to a conflict of interest include a conflict of interest and duty and a conflict of duties; and
  - (iv) an interest of which a director has no knowledge and of which it is unreasonable to expect him or her to have knowledge shall not be treated as an interest of his.

For the avoidance of doubt, where a director ("**first director**") is appointed to act as an alternate by another one or more directors ("**second director**") and the first director has an interest which prevents him or her from voting in relation to any transaction or

arrangement, that first director shall also not be entitled to vote in relation to that transaction or arrangement as alternate on behalf of any second director.

- (e) Subject to article 4.4(f), if a question arises at a meeting of the directors or of a committee of directors as to the right of a director to participate in the meeting (or part of the meeting) for voting or quorum purposes, and that question is not resolved by the director voluntarily agreeing to abstain from voting, the question may, before the conclusion of the meeting, be referred to the Chairperson whose ruling in relation to any director other than the Chairperson is to be final and binding.
- (f) If any question as to the right to participate in a meeting (or part of a meeting) arises in respect of the Chairperson (and that question is not resolved by the Chairperson voluntarily agreeing to abstain from voting) the question is to be decided by a decision of the directors at that meeting, for which purpose the Chairperson is not to be counted as entitled to participate in the meeting (or that part of the meeting) for voting or quorum purposes.

#### 4.5 **Quorum for directors' meetings**

- (a) At a directors' meeting, unless a quorum is participating, no proposal is to be voted on, except a proposal to call another meeting.
- (b) Save as set out in article 4.5(c), the quorum for the transaction of business of the directors shall be two eligible directors.
- (c) The quorum for transaction of business of the directors shall be one eligible director, if:
  - (i) there is a sole director; or
  - (ii) at any meeting of the directors, to the extent called to consider and vote on any matter in relation to which a director is not entitled to or does not vote or whose vote is not counted by virtue of:
    - (A) the provisions of article 4.4(c); or
    - (B) the exercise by a director, pursuant to article 7.1, of the right not to attend and vote; or
    - (C) section 175(6)(b) Companies Act 2006;

there is only one eligible director willing to take a decision on any matter.

(d) If there are no directors in office or the directors in office or the sole director are unable or unwilling to form a quorum or to take a decision on any particular matter, or to appoint further directors to make up a quorum or to enable a decision to be taken on any particular matter, or to call a general meeting or circulate a written resolution to do so, then any shareholder (ignoring for these purposes the Company as the holder of any treasury shares) may call a general meeting or circulate a written resolution or instruct the Company Secretary to do so, for the purposes of taking the decision or appointing one or more additional directors to form a quorum or to enable a decision to be taken.

# 4.6 **Chairing of directors' meetings and chairperson's casting vote**

- (a) The directors may appoint (and remove at any time) a director to chair their meetings and the person so appointed for the time being is known as the Chairperson. If the Chairperson is not participating in a directors' meeting within ten minutes of the time at which it was to start, or is unwilling or unable to act as chairperson at that meeting or any part of it, the participating directors must appoint one of themselves who is willing and able so to act, to be the Chairperson for that meeting or for that part of the meeting.
- (b) If the numbers of votes for and against a proposal are equal, the Chairperson or other director chairing the meeting has a casting vote, unless in relation to a particular proposal at a meeting, the Chairperson or other director chairing the meeting is not an eligible director.

# 4.7 **Records of decisions to be kept**

The directors must ensure that the Company keeps a permanent record in writing which can be read by the naked eye, for at least 10 years from the date of the decision recorded, of every unanimous or majority decision taken by the directors and of any decisions taken by a sole director.

# 4.8 Directors' discretion to make further rules

Subject to these articles and the Companies Act 2006, the directors may make any rule which they think fit about how they take decisions, and about how such rules are to be recorded or communicated to directors.

# 5. **DIRECTORS' PERMITTED INTERESTS**

- 5.1 Provided that (a) he or she has declared the nature and extent of his or hers interest in accordance with (and to the extent required by) the provisions of article 5.4; and (b) the directors or the shareholders have not (upon request) refused to give specific authorisation pursuant to article 6 for a particular situation or matter; and (c) the directors and shareholders have not otherwise resolved pursuant to article 6.3 that a particular situation or matter shall no longer be authorised; a director, notwithstanding his or hers office, shall be authorised:
  - (a) to enter into, or otherwise be interested in, any transaction or arrangement with the Company or any other Group Company or in which the Company (or any other Group Company) is interested, either with regard to his or hers tenure of any office or position in the management, administration or conduct of its business or as seller, buyer or otherwise;
  - (b) to hold any office or place of profit (except that of auditor) with, or to be employed by or a consultant to or otherwise interested (including by way of the holding of shares or securities convertible into shares) in, the Company, any other Group Company or in

any shareholder holding a majority of the voting rights attaching to the issued share capital of the Company or any Associate of any such shareholder;

- (c) to act by himself or by any firm of which he or she is a partner, director, employee or member in a professional capacity (except as auditor) for the Company, any other Group Company or any shareholder holding a majority of the voting rights attaching to the issued share capital of the Company or any Associate of any such shareholder and he or she or his or hers firm shall be entitled to remuneration for professional services as if he or she were not a director of the Company; and
- (d) to be a director of any other company in which the Company does not have an interest if that cannot reasonably be regarded as likely to give rise to a conflict of interest at the time of his or hers appointment as a director of the Company or that other company (whichever is the later),

and such authorisations shall extend to any direct or indirect interest that conflicts or possibly may conflict with the interests of the Company which may reasonably be expected to arise out of the situations and matters so authorised and which is capable of being authorised at law. No authorisation shall be required pursuant to article 6 of any such situation or matter authorised by this article 5.1 and, without limitation, no director shall, by reason of his or hers holding office as a director of the Company (or of the fiduciary relationship established by his or hers holding that office) be liable to account to the Company for any remuneration, profit or other benefit received as a result of any interest permitted by this article 5.1 and no transaction or arrangement shall be liable to be avoided by reason of any director having any interest or having received any benefit permitted by this article 5.1.

- 5.2 The authorisations given pursuant to and the other provisions of article 5.1 shall extend to and include, without limitation, direct or indirect interests of a director which arise (or which may potentially arise) due to:
  - (a) any transaction entered into by the director or any shareholder holding the majority of the voting rights attaching to the issued share capital of the Company or any Associate of that shareholder in relation to shares (or securities convertible into shares) debentures or other securities in (a) the Company or any other Group Company; or in (b) such shareholder or in any such Associate of such shareholder;
  - (b) any guarantee, security or indemnity given or proposed to be given by any Group Company to, or to any person for the benefit of, (a) any other Group Company; or (b) any shareholder holding the majority of the voting rights attaching to the issued share capital of the Company or any Associate of that shareholder;
  - (c) the recommendation, declaration and payment of any dividend or other distribution by the Company;
  - (d) any transaction or arrangement proposed, made, terminated or varied between (a) the Company and any other Group Company; or (b) the Company and any shareholder holding the majority of the voting rights attaching to the issued share capital of the Company or any Associate of that shareholder including without limitation transactions

or arrangements relating to the sale and supply of goods and services, the borrowing or advancing of money and the use of property and other assets; and

(e) any claim or right arising between (a) the Company and any other Group Company; or
(b) the Company and any shareholder holding the majority of the voting rights attaching to the issued share capital of the Company or any Associate of that shareholder.

It shall be a term and condition of the authorisation given pursuant to article (e) that the director shall not be entitled to vote or participate in any discussions relating to the exercise, enforcement or pursuance of any claim or right so authorised.

- 5.3 For the purposes of articles 5.1 and 5.2:
  - (a) an interest of: (a) a person who is connected with a director (within the meaning of section 252 of the Companies Act 2006); and (b) the Appointor in relation to any alternate; shall be treated as an interest of the director or the alternate (as appropriate) in each case in addition to any interest which the director or alternate otherwise has; and
  - (b) any authorisation of a situation or matter pursuant to articles 5.1 and 5.2 relating to a Group Company, or to any shareholder holding the majority of the voting rights in the share capital of the Company or any Associate of that shareholder, shall be effective only for so long as the relevant Group Company remains a Group Company, the relevant shareholder holds the majority of the voting rights in the Company and the relevant Associate remains an Associate of a person who holds the majority of the voting rights in the Company.
- 5.4 In relation to transactions or arrangements with the Company, the director shall declare the nature and extent of any interest authorised under articles 5.1 and 5.2 in any way permitted by the Companies Act 2006 and shall only be required to make such disclosure to the extent required to do so under the Companies Act 2006. In relation to other situations of actual or potential conflict of interest, the director shall declare the nature and extent of his or hers interest at a meeting of the directors, or as otherwise determined by the directors, but shall not be required to declare the nature and extent of his or hers interest to the extent that the other directors are already aware of the interest and its extent.
- 5.5 For the purposes of this article 5, when calculating whether any shareholder holds a majority of the voting rights attaching to the issued share capital of the Company, the voting rights attaching to any shares held by the Company as treasury shares shall be ignored.

## 6. AUTHORISATION OF CONFLICTS OF INTEREST

- 6.1 Any Relevant Matter may be authorised by the directors to the fullest extent permitted by law in accordance with the provisions of articles 6.2 to 6.4.
- 6.2 Any director may propose that a Relevant Matter be authorised by the directors. Such proposal and any authorisation given by the directors shall be effected in the same way as any other matter may be proposed to, and resolved upon by, the directors in accordance with these articles (or in such other manner as all the directors may approve), except that no authorisation

shall be effective unless the requirements of section 175(6) of the Companies Act 2006 have been complied with. Any authorisation of a matter pursuant to this article 6 shall, unless it states otherwise, extend to any actual or potential conflict of interest which may reasonably be expected to arise out of the matter so authorised.

- 6.3 Any authorisation of a matter under this article 6 shall be subject to such terms, conditions and limitations as the directors may specify, whether at the time of giving the authorisation or subsequently. The directors or the shareholders may terminate or vary (including by imposing new terms, conditions and limitations in relation to) any authorisation given under this article 6 or under article 5.1 for the purpose of section 175 of the Companies Act 2006 at any time, but no such termination or variation shall be of retrospective effect. The director concerned must act in accordance with any terms, conditions or limitations specified by the directors or the shareholders in accordance with this article 6.3.
- 6.4 No director shall, by reason of his or hers office as director of the Company (or by reason of the fiduciary relationship established by holding that office), be liable to account to the Company for any benefit derived from any Relevant Matter to the extent that the Relevant Matter has been authorised by the directors in accordance with this article 6. No transaction or arrangement shall be liable to be avoided by reason of any interest of a director to the extent that it has been so authorised.
- 6.5 Notwithstanding the other provisions of this article 6, the shareholders of the Company shall be entitled to authorise a Relevant Matter (whether or not authorisation has previously been requested from and/or refused by the directors) and any authorisation of a matter pursuant to this article 6.5 shall, unless it states otherwise, extend to any actual or potential conflict of interest which may reasonably be expected to arise out of the matter so authorised. The provisions of articles 6.3 and 6.4 shall apply mutatis mutandis to any authorisation so given by the shareholders save that the word(s) "directors" or "directors or shareholders" when referring to the authorisation being given or to any terms and conditions of authorisation being specified, imposed, varied or terminated shall be read only as the word "shareholders". Any authorisation, and the variation or termination of any authorisation by the shareholders under article 6.3 or this article 6.5 shall be by ordinary resolution, save where any greater majority is otherwise required by the Act or other applicable law.

## 7. DIRECTORS' INTERESTS GENERAL

- 7.1 Where this article 7.1 applies, a director shall be deemed to have the authority, without breaching the general duties he or she owes to the Company by virtue of sections 171 to 177 of the Companies Act 2006 to take (and shall take if so requested by the other directors or by the shareholders) such steps as may be necessary or desirable for the purpose of managing any conflict of interest to which this article 7.1 applies, including (without limitation) by:
  - (a) complying with any procedures laid down from time to time by the directors or shareholders for the purpose of managing conflicts of interest generally or any specific procedures approved by the directors or shareholders in relation to the situation, matter or interest in question;
  - (b) excluding himself from attending and voting at board meetings or otherwise participating in directors' decision making to the extent relating to such situation, matter

or interest or from participating in discussions (whether at meetings of the board or otherwise), or receiving documents or information to the extent relating to any such situation, matter or interest (including without limitation, notice of meetings, board papers, minutes or draft minutes, directors' written resolutions and legal advice given to any Group Company);

- (c) arranging for documents or information relating to any such situation, matter or interest to be reviewed by a professional adviser to ascertain the extent to which it might be appropriate for him or her to have access to such documents or information; and/or
- (d) not disclosing to the Company, or not using in relation to the Company's affairs, information which he or she obtains or has obtained otherwise than through his or hers position as a director of the Company which relates to a situation, matter or interest and which is confidential to a third party, where to do so would amount to a breach of confidence or breach of duty to the third party.
- 7.2 Article 7.1 shall apply, where a director has or could have:
  - (a) a direct or indirect interest that conflicts or possibly may conflict with the interests of the Company and provided that the interest or the existence of the situation or relationship leading to the interest has been authorised pursuant to article 5.1 or article 6 and unless otherwise specified by the terms and conditions of such authorisation; and
  - (b) a direct or indirect interest in a transaction or arrangement with the Company and such interest has been declared to the other directors to the extent required by the Companies Act 2006.
- 7.3 Where a director obtains or has obtained information, otherwise than through his or hers position as a director, which is confidential to a third party other than the Company, then provided that the duty of confidentiality does not arise out of a situation in which the director has or may have a direct or indirect conflict of interest, the director shall not be required to disclose such information to the Company or use it in relation to the Company's affairs. This article is without prejudice to the ability of a director to withhold such information from the Company in accordance with the provisions of article 7.1.
- 7.4 Articles 7.1 and 7.3 are without prejudice to any equitable principle or rule of law which may otherwise excuse or release the director from any requirement to disclose information or use information in relation to the Company's affairs, participate in discussions or receive documents or information.
- 7.5 For the purposes of articles 5 to 7 references to a conflict of interest include a conflict of interest and duty and a conflict of duties.

# 8. ALTERNATE DIRECTORS

8.1 Any director, other than an alternate director (**"Appointor**") may appoint as an alternate any other director, or any other person who is willing to act, to exercise (in the absence of the Appointor) the Appointor's powers as a director generally, and in particular but without limitation

(in the absence of the Appointor) to carry out the Appointor's responsibilities in relation to the taking of decisions by directors.

- 8.2 Any appointment or removal of an alternate must be effected by notice in writing to the Company signed by the Appointor, or in any other manner approved by the directors. The notice must identify the proposed alternate and, in the case of a notice of appointment, contain a statement signed by the proposed alternate that the proposed alternate is willing to act as the alternate of the director giving the notice.
- 8.3 An alternate director has the same rights, in relation to any directors' meeting or decision of the directors, as the alternate's Appointor and, without limitation, is entitled to receive notice of all meetings of directors and committees of directors and all meetings of shareholders which their Appointor is entitled to receive and in the absence of their Appointor, to attend, speak and vote at all such meetings at which the Appointor is entitled to attend, speak and vote.
- 8.4 Except as these articles specify otherwise, alternate directors are deemed for all purposes to be directors, are liable for their own acts and omissions, are subject to the same restrictions as their Appointors, and are not deemed to be agents of or for their Appointors.
- 8.5 Subject to article 8.6, a person who is an alternate director, but not a director:
  - (a) may be counted as participating in a directors' meeting for the purposes of determining whether a quorum is present and may vote on any proposal made at a directors' meeting (if that person's Appointor is not participating but would have been an eligible director in relation to that proposal had he or she been participating); and
  - (b) may take part in decisions of the directors pursuant to article 4.2 (provided that person's Appointor does not take part in making the decision but would have been an eligible director in relation to that decision had he or she taken part in making it).
- 8.6 A person may be appointed as the alternate director of more than one director. Where a person is appointed as the alternate director of more than one director, or is an alternate director and a director himself, that alternate director shall (subject to article 4.4):
  - (a) be entitled at meetings of the directors to one vote in respect of every director by whom he or she has been appointed (and who is not himself participating, but who would have been an eligible director in relation to the proposal had he or she been participating) in addition to his or hers own vote (if any) as a director;
  - (b) may be counted more than once for the purpose of determining whether or not a quorum is present; and
  - (c) shall be entitled to take part in decisions of the directors pursuant to article 4.2 on behalf of each director by whom he or she has been appointed (and who would have been an eligible director in relation that decision) as well as being able to take part in making the decision for himself (if he or she is a director).
- 8.7 An alternate director is not entitled to receive any remuneration from the Company for serving as an alternate director except such part of the remuneration of the alternate's Appointor as the Appointor may direct by notice in writing made to the Company.

- 8.8 An alternate director's appointment as an alternate for a particular Appointor shall terminate:
  - (a) when that Appointor revokes the appointment by notice to the Company in writing specifying when it is to terminate;
  - (b) on the death of that Appointor; or
  - (c) when the directorship of that Appointor terminates;

and an alternate director's appointment as an alternate for an Appointor (and, if the person is an alternate for more than one director, that person's appointment as an alternate for each Appointor) shall terminate on the occurrence in relation to the alternate of any event which, if it occurred in relation to any Appointor of that alternate, would result in the termination of that Appointor's appointment as a director.

## 9. **APPOINTMENT, RETIREMENT AND REMOVAL OF DIRECTORS**

- 9.1 Any person who is willing to act as a director, and who is permitted by law to do so, may be appointed to be a director by ordinary resolution, or by a decision of the directors.
- 9.2 In any case where, as a result of death or bankruptcy, the Company has no shareholders (ignoring for these purposes the Company as holder of any treasury shares) and no directors, the Transmittee(s) of the last shareholder to have died or to have had a bankruptcy order made against him or her (as the case may be) have the right, by notice in writing, to appoint a natural person who is willing to do so to be a director and any such appointment shall be as effective as if made by the Company in general meeting pursuant to these articles. For the purposes of this article, where two or more shareholders die in circumstances rendering it uncertain who was the last to die, a younger shareholder is deemed to have survived an older shareholder.
- 9.3 A person ceases to be a director as soon as:
  - that person ceases to be a director by virtue of any provision of the Companies Act 2006 or is prohibited from being a director by law;
  - (b) (in case of a director who is a natural person) a bankruptcy order is made against that person or a composition is made with that person's creditors generally in satisfaction of that person's debts;
  - (c) (in the case of a director which is a body corporate) that body corporate (i) passes any resolution for voluntary winding up (within the meaning of section 84(2) Insolvency Act 1986 or otherwise) or is wound up by the court; (ii) is the subject of an administration order or an administrator is appointed in respect of that body corporate; (iii) makes any proposal under Part I Insolvency Act 1986 or otherwise for a composition in satisfaction of its debts or a scheme of arrangement of its affairs or makes any proposal under part 26 Companies Act 2006 or otherwise for a compromise or arrangement between it and its creditors or any class of them, makes any arrangement or compromise with creditors generally or ceases to carry on all or substantially all of its business; (iv) has an administrative receiver, receiver or manager appointed over all or any substantial part

of its assets; or is the subject of any occurrence substantially similar in nature or effect, whether in England and Wales or any other jurisdiction;

- (d) a registered medical practitioner who is treating that person gives a written opinion to the Company stating that that person has become physically or mentally incapable of acting as a director and may remain so for more than three months;
- (e) (where the director has not participated in decision making of the directors for more than six months and the directors believe this to be by virtue of any mental or physical incapacity of the director) the directors resolve that his or hers office be vacated; or
- (f) notification is received by the Company from the director that the director is resigning from office as director and such resignation has taken effect in accordance with its terms.

#### 10. DIRECTORS' REMUNERATION AND EXPENSES

- 10.1 Directors may undertake any services for the Company that the directors decide and shall be entitled to such remuneration in such form as the directors determine both for their services to the Company as directors and for any other service which they undertake for the Company. Unless the directors decide otherwise such remuneration shall accrue from day to day and directors shall not be accountable to the Company for any remuneration which they receive as directors or other officers or employees of the Company's subsidiaries or of any other body corporate in which the Company is interested.
- 10.2 The Company may pay any reasonable expenses which the directors (including any alternate director) or the Company Secretary properly incur in connection with their attendance at meetings of directors or committees of directors, general meetings, or separate meetings of the holders of any class of shares or of debentures of the Company, or otherwise in connection with the proper exercise of their powers and the discharge of their responsibility in relation to the Company.
- 10.3 The directors may exercise all the powers of the Company to provide benefits, whether by the payment of gratuities or pensions or by insurance or otherwise, for any director or former director who holds or has held any executive office or employment with the Company or with any body corporate which is or has been a Group Company or with a predecessor in business of the Company or of any such body corporate, and for any member of his or hers family (including a spouse, former spouse, civil partner or former civil partner) or any person who is or was dependent on him, and may contribute to any fund and pay premiums for the purchase or provision of any such benefit.

#### 11. SHARES: GENERAL

#### 11.1 Shares to be fully paid up

All shares shall be issued fully paid.

#### 11.2 **Power to issue different classes of share**

- (a) Subject to these articles, but without prejudice to the rights attached to any existing share, the Company may issue shares with such rights or restrictions as may be determined by ordinary resolution.
- (b) The Company may issue shares which are to be redeemed, or are liable to be redeemed at the option of the Company or the relevant shareholder, and the directors may determine the terms, conditions and manner of redemption of any such shares.

## 11.3 **Disapplication of pre-emption on allotment**

In accordance with section 567(1) Companies Act 2006, section 561 and 562 Companies Act 2006 shall not apply to any allotment of equity securities made by the Company.

#### 11.4 Absolute interests only

Except as required by law, no person is to be recognised by the Company as holding any share upon any trust, and except as otherwise required by law or these articles, the Company is not in any way to be bound by or recognise any interest in a share other than the shareholder's absolute ownership of it and all the rights attaching to it.

## 12. UNCERTIFICATED SHARES

- 12.1 Under and subject to the uncertificated securities rules, the board may permit title to shares of any class to be evidenced otherwise than by certificate and title to shares of such a class to be transferred by means of a relevant system and may make arrangements for a class of shares (if all shares of that class are in all respects identical) to become a participating class. Title to shares of a particular class may only be evidenced otherwise than by a certificate where that class of shares is at the relevant time a participating class. The board may also, subject to compliance with the uncertificated securities rules, determine at any time that title to any class of shares may from a date specified by the board no longer be evidenced otherwise than by a certificate or that title to such a class shall cease to be transferred by means of any particular relevant system.
- 12.2 In relation to a class of shares which is a participating class and for so long as it remains a participating class, no provision of these articles shall apply or have effect to the extent that it is inconsistent in any respect with:
  - (a) the holding of shares of that class in uncertificated form;
  - (b) the transfer of title to shares of that class by means of a relevant system; or
  - (c) any provision of the uncertificated securities rules,

and, without prejudice to the generality of this article, no provision of these articles shall apply or have effect to the extent that it is in any respect inconsistent with the maintenance, keeping or entering up by the Operator, so long as that is permitted or required by the uncertificated securities rules, of an Operator register of securities in respect of that class of shares in uncertificated form.

- 12.3 Shares of a class which is at the relevant time a participating class may be changed from uncertificated to certificated form, and from certificated to uncertificated form, in accordance with and subject as provided in the uncertificated securities rules.
- 12.4 If, under these articles or the Companies Acts, the Company is entitled to sell, transfer or otherwise dispose of, forfeit, re-allot, accept the surrender of or otherwise enforce a lien over an uncertificated share, then, subject to these articles and the Companies Acts, such entitlement shall include the right of the board to:
  - (a) require the holder of the uncertificated share by notice in writing to change that share from uncertificated to certificated form within such period as may be specified in the notice and keep it as a certificated share for as long as the board requires;
  - (b) appoint any person to take such other steps, by instruction given by means of a relevant system or otherwise, in the name of the holder of such share as may be required to effect the transfer of such share and such steps shall be as effective as if they had been taken by the registered holder of that share; and
  - (c) take such other action that the board considers appropriate to achieve the sale, transfer, disposal, forfeiture, re-allotment or surrender of that share or otherwise to enforce a lien in respect of that share.
- 12.5 Unless the board determines otherwise, shares which a member holds in uncertificated form shall be treated as separate holdings from any shares which that member holds in certificated form but a class of shares shall not be treated as two classes simply because some shares of that class are held in certificated form and others in uncertificated form.
- 12.6 Unless the board determines otherwise or the uncertificated securities rules require otherwise, any shares issued or created out of or in respect of any uncertificated shares shall be uncertificated shares and any shares issued or created out of or in respect of any certificated shares shall be certificated shares.
- 12.7 The Company shall be entitled to assume that the entries on any record of securities maintained by it in accordance with the uncertificated securities rules and regularly reconciled with the relevant Operator register of securities are a complete and accurate reproduction of the particulars entered in the Operator register of securities and shall accordingly not be liable in respect of any act or thing done or omitted to be done by or on behalf of the Company in reliance on such assumption. Any provision of these articles which requires or envisages that action will be taken in reliance on information contained in the Register shall be construed to permit that action to be taken in reliance on information contained in any relevant record of securities (as so maintained and reconciled).

## 13. SHARE CERTIFICATES

13.1 Every person (except a person to whom the Company is not by law required to issue a certificate) whose name is entered in the Register as a holder of any certificated shares shall be entitled, without charge, to receive within the time limits prescribed by the Companies Acts (unless the terms of issue prescribe otherwise) one certificate for all of the shares of that class registered in his or her name.

- 13.2 The Company shall not be bound to issue more than one certificate in respect of shares held jointly by two or more persons. Delivery of a certificate to the person first named in the Register shall be sufficient delivery to all joint holders.
- 13.3 Where a member has transferred part only of the shares comprised in a certificate, he or she shall be entitled without charge to a certificate for the balance of such shares to the extent that the balance is to be held in certificated form. Where a member receives more shares of any class, he or she shall be entitled without charge to a certificate for the extra shares of that class to the extent that the balance is to be held in certificate form.
- 13.4 A share certificate may be issued under seal (by affixing the seal to or printing the seal or a representation of it on the certificate) or signed by at least two Directors (or by at least one Director and the Company Secretary, if appointed). Such certificate shall specify the number and class of the shares in respect of which it is issued and the amount or respective amounts paid up on it. The board may by resolution decide, either generally or in any particular case or cases, that any signatures on any share certificates need not be autographic but may be applied to the certificates by some mechanical or other means or may be printed on them or that the certificates need not be signed by any person.
- 13.5 Every share certificate sent in accordance with these articles will be sent at the risk of the member or other person entitled to the certificate. The Company will not be responsible for any share certificate lost or delayed in the course of delivery.

# 14. **REPLACEMENT CERTIFICATES**

- 14.1 Any two or more certificates representing shares of any one class held by any member may at his or her request be cancelled and a single new certificate for such shares issued in lieu without charge on surrender of the original certificates for cancellation.
- 14.2 Any certificate representing shares of any one class held by any member may at his or her request be cancelled and two or more certificates for such shares may be issued instead.
- 14.3 If a share certificate is defaced, worn out or said to be stolen, lost or destroyed, it may be replaced on such terms as to evidence and indemnity in respect of such share certificate only as the board may decide and, where it is defaced or worn out, after delivery of the old certificate to the Company.
- 14.4 The board may require the payment of any exceptional out-of-pocket expenses of the Company incurred in connection with the issue of any certificates under this article. In the case of shares held jointly by several persons, any such request as is mentioned in this article may be made by any one of the joint holders.

## 15. SHARES: TRANSFER

## 15.1 General

(a) Each member may transfer all or any of his or her shares which are in certificated form by instrument of transfer in writing in any usual form or in any form approved by the board. Such instrument shall be executed by or on behalf of the transferor and (in the case of a transfer of a share which is not fully paid up) by or on behalf of the transferee. All instruments of transfer, when registered, may be retained by the Company.

- (b) Each member may transfer all or any of his or her shares which are in uncertificated form by means of a relevant system in such manner provided for, and subject as provided in, the uncertificated securities rules. No provision of these articles shall apply in respect of an uncertificated share to the extent that it requires or contemplates the effecting of a transfer by an instrument in writing or the production of a certificate for the share to be transferred.
- (c) The transferor remains the holder of a share until the transferee's name is entered in the register of members as shareholder in respect of it.
- (d) Subject to these articles (in particular, but without limitation, article 15.2(a)) the directors may, in their absolute discretion, refuse to register the transfer of a share in certificated form. If they do so, then as soon as practicable and in any event within two months after the date on which the transfer is lodged with the Company, the instrument of transfer must be returned to the transferee with the notice of refusal together with reasons for such refusal, unless they suspect that the proposed transfer may be fraudulent.
- (e) The directors may refuse to register a transfer of uncertificated shares in any circumstances that are allowed or required by the uncertificated securities rules and the relevant system.

# 15.2 Transfer of shares and pre-emption on transfer in relation to security held by a secured institution

- (a) Notwithstanding anything contained in these articles, the directors shall not decline to register any transfer of shares, nor may they suspend registration of any transfer of shares where such transfer:
  - (i) is to any bank or institution or other person to which such shares have been charged or mortgaged, or to any nominee of such a bank or institution or other person ("Secured Institution"); or
  - (ii) is delivered to the Company for registration by a Secured Institution or its nominee in order to perfect its security over the shares; or
  - (iii) is executed by a Secured Institution or its nominee pursuant to the power of sale or other power under its security over the shares,

and the directors shall register any such transfer of shares forthwith following receipt.

(b) Notwithstanding anything to the contrary contained in these articles, no transferor or proposed transferor of any shares in the Company to a Secured Institution or its nominee and no Secured Institution or its nominee shall be required to offer the shares which are or are to be the subject of any transfer referred to in articles 15.2(a)(i) to 15.2(a)(iii) inclusive to the shareholders for the time being of the Company or any of them, and no such shareholder shall have any right under these articles or under any

agreement or otherwise to require those shares to be offered to or transferred to it whether for consideration or not.

## 16. SHARES: DRAG ALONG

- 16.1 If any one or more members receives an offer in writing from a bona fide third party ("**Third Party**") to purchase the entire equity share capital in the Company not already owned by the Third Party ("**Third Party Offer**") and the holders of at least 75% of the issued shares accept the Third Party Offer ("**Accepting Shareholders**"), the Accepting Shareholders are entitled to issue to the remaining members ("**Other Shareholders**") written notice ("**Drag Along Notice**") requiring the Other Shareholders to sell to the Third Party all of the Other Shareholders' shares upon the terms and conditions specified in the Drag Along Notice.
- 16.2 The terms on which the Accepting Shareholders require the Other Shareholders to sell their shares must be no less favourable than the terms on which the Accepting Shareholders are selling their shares to the Third Party.
- 16.3 The Drag Along Notice must specify:
  - (a) the details of the Third Party;
  - (b) the price payable for each share and other consideration (if any) to be received (directly or indirectly) by the Accepting Shareholders; and
  - (c) any other material terms upon which the Other Shareholders' shares shall be purchased pursuant to the Drag Along Notice.
- 16.4 If any Other Shareholder shall not, within five Business Days of being required to do so, execute and deliver transfers in respect of the shares held by such Other Shareholder and deliver the certificate(s) in respect of the same (or a suitable indemnity in lieu thereof), then any Accepting Shareholder shall be entitled to execute, and shall be entitled to authorise and instruct such person as the relevant Accepting Shareholder thinks fit to execute, the necessary transfer(s) and indemnities on the Other Shareholder's behalf and, against receipt by the Company (on trust for such Shareholder) of the consideration payable for the relevant shares, deliver such transfer(s) and certificate(s) or indemnities to the Third Party (or their nominee) and register such Third Party (or their nominee) as the holder thereof and, after such registration, the validity of such proceedings shall not be questioned by any person.
- 16.5 The Other Shareholders are not obliged to sell their shares in accordance with this article 16 if the Accepting Shareholders do not complete the sale of all their shares to the Third Party on the same terms and conditions set out in the Drag Along Notice.

# 17. SHARES: TAG ALONG

17.1 If at any time one or more members ("**Proposed Sellers**") propose to sell to any person ("**Proposed Buyer**"), in one or a series of related transactions, such number of shares which when registered would result in that person (together with persons connected or acting in concert with such person) holding or increasing their holding to 50% or more of the issued equity share capital of the Company ("**Proposed Sale**"), the Proposed Sellers shall give written

notice ("**Tag Along Notice**") to the other holders of shares of the Proposed Sale at least 10 Business Days prior to the proposed date of completion thereof.

- 17.2 The Tag Along Notice must specify:
  - (a) the details of the Proposed Buyer;
  - (b) the sale price for each share and other consideration (if any) to be received (directly or indirectly) by the Proposed Sellers; and
  - (c) any other material terms upon which the shares are to be purchased.
- 17.3 The Proposed Sale may not be completed unless the Proposed Buyer has unconditionally offered to buy all the other issued shares (other than any shares already owned by the Proposed Buyer or persons connected or acting in concert with such Proposed Buyer) on the same terms and conditions as apply to the Proposed Sale. Such offer shall remain open for acceptance for not less than 10 days.
- 17.4 The provisions of this article 17 shall not apply to any Proposed Sale which is to take place pursuant to a Third Party Offer under article 16.

## 18. SHARES: TRANSMISSION

- 18.1 If title to a share passes to a Transmittee, the Company may recognise only the Transmittee as having any title to that share. Subject to these articles, a Transmittee who produces such evidence of entitlement to shares as the directors may properly require may choose either to become the shareholder of those shares (and for the avoidance of doubt, article 15.1(d) shall not apply in such circumstances) or (subject to article 15.1(d)) to have them transferred to another person, and subject to article 18.2 pending any transfer of the shares to another person, has the same rights as the shareholder had.
- 18.2 Subject to article 9.2, Transmittees do not have the right to attend or vote at a general meeting or to agree to a proposed written resolution, in respect of shares to which they are entitled by reason of a shareholder's death or bankruptcy or otherwise, unless they become the shareholders of those shares.
- 18.3 Transmittees who wish to become shareholders in relation to shares to which they have become entitled must notify the Company in writing of that wish. Transmittees who wish to have a share transferred to another person must execute an instrument of transfer in respect of it and any transfer made or executed under this article is to be treated as if it were made or executed by the person from whom the Transmittee has derived rights in respect of the share, and as if the event which gave rise to the transmission had not occurred.
- 18.4 If a notice is given to a shareholder in respect of shares and a Transmittee is entitled to those shares, the Transmittee is bound by the notice if it was given to the shareholder before the Transmittee's name or the name of any person(s) named as the transferee(s) in an instrument of transfer executed under article 18.3, has been entered in the register of members.
- 18.5 Transmittees entitled to a share in uncertificated form who elect to have some other person registered shall either:

- (a) procure that instructions are given by means of the relevant system to effect transfer of such uncertificated share to that person; or
- (b) change the uncertificated share to certificated form and execute an instrument of transfer of that certificated share to that person.

# 19. DIVIDENDS AND OTHER DISTRIBUTIONS

#### 19.1 **Procedure for declaring dividends**

- (a) The Company may by ordinary resolution declare dividends, and the directors may decide to pay interim dividends.
- (b) No dividend may be declared or paid unless it is in accordance with shareholders' respective rights. Unless the shareholders' resolution to declare or the directors' decision to pay a dividend, or the terms on which shares are issued, specify otherwise, it must be paid by reference to each shareholder's holding of shares on the date of the resolution or decision to declare or pay it. Any shares held by the Company as treasury shares shall be ignored for the purposes of calculating each shareholder's entitlement to any dividend or distribution.
- (c) The directors may pay at intervals any dividend payable at a fixed rate if it appears to them that the profits available for distribution justify the payment.

## 19.2 **Payment of dividends and other distributions**

- (a) Where a dividend or other sum which is a distribution is payable in respect of a share, it must be paid by one or more of the following means:
  - transfer to a bank or building society account specified by the Distribution Recipient either in writing or as the directors may otherwise decide;
  - (ii) sending a cheque made payable to the Distribution Recipient by post (in accordance with article 27.4) to the Distribution Recipient at the Distribution Recipient's registered address (if the Distribution Recipient is the shareholder of the share), or (in any other case) to an address specified by the Distribution Recipient in writing or (where no such address has been specified) as the directors may otherwise decide;
  - (iii) sending a cheque made payable to such person by post (in accordance with article 27.4) to such person at such address as the Distribution Recipient has specified in writing or (where no such address has been specified) as the directors may otherwise decide; or
  - (iv) any other means of payment as the directors agree with the Distribution Recipient in writing.
- (b) For uncertificated shares, any payment may be made by means of the relevant system (subject always to the facilities and requirements of the relevant system) and such payment may be made by the Company or any person on its behalf by sending an

instruction to the operator of the relevant system to credit the cash memorandum account of the Distribution Recipient of such shares or, if permitted by the Company, of such person as the Distribution Recipient may in writing direct.

- (c) In these articles, "**Distribution Recipient**" means, in respect of a share in respect of which a dividend or other sum is payable:
  - (i) the shareholder of the share (ignoring for these purposes the Company as holder of any treasury share); or
  - (ii) if the share has two or more joint shareholders, whichever of them is named first in the register of members; or
  - (iii) if the shareholder is no longer entitled to the share by reason of death or bankruptcy, or otherwise by operation of law, the Transmittee.

## 19.3 No interest on distributions

The Company may not pay interest on any dividend or other sum payable in respect of a share unless otherwise provided by the terms on which the share was issued, or the provisions of another agreement between the shareholder of that share and the Company.

# 19.4 Unclaimed distributions

- (a) All dividends or other sums which are payable in respect of shares and which are unclaimed after having been declared or becoming payable, may be invested or otherwise made use of by the directors for the benefit of the Company until claimed. The payment of any such dividend or other sum into a separate account does not make the Company a trustee in respect of it.
- (b) If twelve years have passed from the date on which a dividend or other sum became due for payment and the Distribution Recipient has not claimed it, the Distribution Recipient is no longer entitled to that dividend or other sum and it ceases to remain owing by the Company.

## 19.5 Non-cash distributions

- (a) Subject to the terms of issue of the share in question, the Company may, by ordinary resolution, decide to pay all or part of a dividend or other distribution payable in respect of a share by transferring non-cash assets of equivalent value (including, without limitation, shares or other securities in any company).
- (b) For the purposes of paying a non-cash distribution, the directors may make whatever arrangements they think fit, including (where any difficulty arises regarding the distribution) fixing the value of any assets, paying cash to any Distribution Recipient on the basis of that value in order to adjust the rights of Distribution Recipients, and vesting any assets in trustees.

#### 19.6 Waiver of distributions

Distribution Recipients may waive their entitlement to a dividend or other distribution payable in respect of a share by giving the Company notice in writing to that effect prior to the declaration of a dividend by a general meeting, or the payment of an interim dividend decided on by the directors, but if the share has more than one shareholder, or more than one person is entitled to the share whether by reason of the death or bankruptcy of one or more joint shareholders or otherwise, the notice is not effective unless it is expressed to be given, and signed, by all the shareholders or persons otherwise entitled to the share.

## 20. CAPITALISATION OF PROFITS

- 20.1 Subject to these articles, the directors may, if they are so authorised by an ordinary resolution:
  - (a) decide to capitalise any profits of the Company (whether or not they are available for distribution) which are not required for paying a preferential dividend, or any sum standing to the credit of the Company's share premium account or capital redemption reserve; and
  - (b) appropriate any sum which they so decide to capitalise ("**capitalised sum**") to the persons who would have been entitled to it if it were distributed by way of dividend ("**persons entitled**") and in the same proportions.
- 20.2 Capitalised sums must be applied on behalf of the persons entitled, and in the same proportions as a dividend would have been distributed to them.
- 20.3 Any capitalised sum may be applied in paying up new shares of a nominal amount equal to the capitalised sum which are then allotted credited as fully paid to the persons entitled or as they may direct.
- 20.4 A capitalised sum which was appropriated from profits available for distribution may be applied in paying up new debentures of the Company which are then allotted credited as fully paid to the persons entitled or as they may direct.
- 20.5 Subject to these articles the directors may:
  - (a) apply capitalised sums in accordance with articles 20.3 and 20.4 partly in one way and partly in another;
  - (b) make such arrangements as they think fit to deal with shares or debentures becoming distributable in fractions under this article (including the issuing of fractional certificates or the making of cash payments or ignoring fractions altogether); and
  - (c) authorise any person to enter into an agreement with the Company on behalf of all the persons entitled which is binding on them in respect of the allotment of shares and debentures to them under this article.

## 21. DECISION-MAKING BY SHAREHOLDERS: ORGANISATION OF GENERAL MEETINGS

#### 21.1 Attendance and speaking at general meetings

- (a) A person is able to exercise the right to speak at a general meeting when that person is in a position to communicate orally to all those attending the meeting, during the meeting, any information or opinions which that person has on the business of the meeting.
- (b) A person is able to exercise the right to vote at a general meeting when:
  - (i) that person is able to vote, during the meeting, on resolutions put to the vote at the meeting, and
  - (ii) that person's vote can be taken into account in determining whether or not such resolutions are passed at the same time as the votes of all the other persons attending the meeting.
- (c) The directors may make whatever arrangements they consider appropriate to enable those attending a general meeting to exercise their rights to speak or vote at it.
- (d) In determining attendance at a general meeting, it is immaterial whether any two or more members attending it are in the same place as each other.
- (e) Two or more persons who are not in the same place as each other attend a general meeting if their circumstances are such that if they have (or were to have) rights to speak and vote at that meeting, they are (or would be) able to exercise them. Such a meeting shall be deemed to take place where the largest group of those persons are assembled, or if there is no such group, where the chairperson of the meeting is located.

## 21.2 Quorum for general meetings

Unless the Company has only one shareholder (ignoring for these purposes the Company as holder of any treasury shares) the quorum required at general meetings and adjourned meetings shall be any two qualifying persons present at the meeting unless: (a) each is a qualifying person only because he or she is authorised to act as the representative of a shareholder which is a corporation in relation to the meeting, and both are representatives of the same corporation; or (b) each is a qualifying person only because he or she is appointed as proxy of a shareholder in relation to the meeting, and both are proxies of the same shareholder. If and for so long as the Company has only one shareholder (ignoring for these purposes the Company as holder of any treasury shares) one qualifying person present at the meeting shall be a quorum. No business other than the appointment of the chairperson of the meeting is to be transacted at a general meeting or an adjourned meeting if the persons attending it do not constitute a quorum. For the purposes of this article 21.2 a "qualifying person" means (i) an individual who is a shareholder of the Company; (ii) a person authorised to act as the representative of a corporation who is a shareholder (ignoring for these purposes the Company as the holder of any treasury shares) in relation to the meeting; or (iii) a person appointed as proxy of a shareholder (ignoring for these purposes the Company as the holder of any treasury shares) in relation to the meeting.

## 21.3 Chairing general meetings

If the directors have appointed a Chairperson, the Chairperson shall chair general meetings if present and willing to do so. If the directors have not appointed a Chairperson, or if the Chairperson is unwilling to chair the meeting or is not present within ten minutes of the time at which a meeting was due to start:

- (a) the directors (or director if there is only one) present; or
- (b) (if no directors are present), any qualifying person (or if more than one) a majority of those qualifying persons present and entitled to vote at the meeting,

must appoint a director or qualifying person to chair the meeting, and the appointment of the chairperson of the meeting must be the first business of the meeting. The person chairing a meeting in accordance with this article is referred to as "the chairperson of the meeting".

# 21.4 Attendance and speaking by directors and non-shareholders

Directors may attend and speak at general meetings, whether or not they are shareholders.

# 21.5 Notice deemed received

A shareholder present either in person or by proxy, at any meeting of the Company shall be deemed to have received notice of the meeting and, where relevant, of the purposes for which it was called.

## 21.6 Adjournment

- (a) If the persons attending a general meeting within half an hour of the time at which the meeting was due to start do not constitute a quorum, or if during a meeting a quorum ceases to be present, the chairperson of the meeting must adjourn the meeting, unless it was called at the request of the shareholders, in which case it must be dissolved. The chairperson of the meeting must also adjourn a general meeting if directed to do so by a meeting at which a quorum is present.
- (b) The chairperson of the meeting may adjourn a general meeting at which a quorum is present if the meeting consents to an adjournment, or it appears to the chairperson of the meeting that an adjournment is necessary to protect the safety of any person attending the meeting or to ensure that the business of the meeting is conducted in an orderly manner.
- (c) When adjourning a general meeting, the chairperson of the meeting must either specify the time and place to which it is adjourned or state that it is to continue at a time and place to be fixed by the directors, and have regard to any directions as to the time and place of any adjournment which have been given by the meeting (where that meeting is quorate).
- (d) Save where: (a) the adjournment is of a temporary nature lasting not more than half an hour; (b) the adjourned meeting is to be held in the same place as the original meeting; and (c) the chairperson announces whilst a quorum is present the time at which the adjourned meeting shall start; at least 5 clear days' notice shall be given of every

adjourned meeting. Such notice shall be given to the same persons to whom notice of the Company's general meetings is required to be given and shall specify the time and place of the meeting and the general nature of the business to be conducted. No further notice of an adjourned meeting is required.

(e) No business may be transacted at an adjourned general meeting which could not properly have been transacted at the meeting if the adjournment had not taken place.

# 22. DECISION-MAKING BY SHAREHOLDERS: VOTING AT GENERAL MEETINGS

## 22.1 Voting: General

A resolution put to the vote of a general meeting must be decided on a show of hands unless a poll is duly demanded in accordance with these articles.

#### 22.2 Voting: Proxies

- (a) Subject to article 22.2(b), on a vote on a resolution on a show of hands at a meeting, every proxy present who has been duly appointed by one or more members entitled to vote on the resolution has one vote.
- (b) On a vote on a resolution on a show of hands at a meeting, a proxy has one vote for and one vote against the resolution if the proxy has been duly appointed:
  - by more than one member entitled to vote on the resolution and the proxy has been instructed by one or more of those members to vote for the resolution and by one or more other of those members to vote against the resolution; or
  - (ii) by a member entitled to vote on the resolution (and who holds the shares on behalf of two or more other persons) and the proxy has been instructed by that member to vote for the resolution in relation to some of the shares held by that member and against the resolution in relation to some other of the shares held by that member.
- (c) On a poll taken at a meeting of a company all or any of the voting rights of a member may be exercised by one or more duly appointed proxies.
- (d) Where a member appoints more than one proxy, article 22.2(c) does not authorise the exercise by the proxies taken together of more extensive voting rights than could be exercised by the member in person.

## 22.3 Errors and disputes

No objection may be raised to the qualification of any person voting at a general meeting except at the meeting or adjourned meeting at which the vote objected to is tendered, and every vote not disallowed at the meeting is valid. Any such objection must be referred to the chairperson of the meeting, whose decision is final.

#### 22.4 Poll Votes

- (a) A poll on a resolution may be demanded in advance of the general meeting where it is to be put to the vote, or at a general meeting, either before a show of hands on that resolution or immediately after the result of a show of hands on that resolution is declared. Unless the chairperson of the meeting determines it would be impractical or unfair to do so, polls must be taken immediately and shall be taken in such manner as the chairperson of the meeting directs
- (b) A poll may be demanded by:
  - (i) the chairperson of the meeting;
  - (ii) the directors;
  - (iii) two or more persons having the right to vote on the resolution; or
  - (iv) a person or persons representing not less than one tenth of the total voting rights of all the shareholders having the right to vote on the resolution (excluding for the avoidance of doubt the voting rights attached to any shares held by the Company as treasury shares).
- (c) A demand for a poll may be withdrawn if the poll has not yet been taken, and the chairperson of the meeting consents to the withdrawal. A demand so withdrawn shall not be taken to have invalidated the result of a show of hands declared before the demand was made.

#### 22.5 **Content of proxy notices**

- (a) Proxies may only validly be appointed by a notice in writing ("**Proxy Notice**") which:
  - (i) states the name and address of the shareholder appointing the proxy;
  - (ii) identifies the person appointed to be that shareholder's proxy and the general meeting in relation to which that person is appointed;
  - (iii) where the proxy is not entitled to exercise the rights attaching to all of the shares held by that shareholder, identifies the number of shares in relation to which the proxy is entitled to exercise such rights;
  - (iv) is signed by or on behalf of the shareholder appointing the proxy, or is authenticated in such manner as the directors may determine; and
  - (v) is delivered to the Company in accordance with these articles and any instructions contained in the notice of the general meeting to which they relate.

Only one proxy may be appointed in any Proxy Notice and a shareholder wishing to appoint more than one proxy must use a separate Proxy Notice for each appointment.

(b) The directors may require Proxy Notices to be delivered in a particular form, and may specify different forms for different purposes. Proxy Notices may specify how the proxy appointed under them is to vote (or that the proxy is to abstain from voting) on one or

more resolutions or may give the proxy discretion as to how to vote on one or more resolutions.

- (c) Unless a Proxy Notice indicates otherwise, it must be treated as:
  - allowing the person appointed under it as a proxy discretion as to how to vote on any ancillary or procedural resolutions put to the meeting;
  - (ii) appointing that person as a proxy in relation to any adjournment of the general meeting to which it relates as well as the meeting itself; and
  - (iii) allowing the person appointed under it as a proxy to exercise the rights attaching to all of the shares held by the shareholder appointing that person as a proxy and no person shall be entitled to challenge the validity of the exercise by such proxy of all those rights.

## 22.6 Delivery of proxy notices

- (a) A Proxy Notice and any authority under which it is signed or otherwise authenticated in such a manner as the directors may determine (or a copy of such authority or other authentication certified notarially or by a solicitor or in some other way approved by the directors) may:
  - (i) in the case of a Proxy Notice in hard copy form, be deposited at the registered office (or at such other place within the United Kingdom as is specified in the notice convening the meeting or in any form of proxy sent out by the Company in relation to the meeting) at any time before the time for holding the meeting or adjourned meeting at which the person named in the appointment proposes to vote; or
    - (A) in the case of a Proxy Notice sent by electronic means where an address has been given by the Company:
    - (B) in the notice calling the meeting; or
    - (C) in any form of proxy sent out by the Company in relation to the meeting; or
    - (D) in any invitation to appoint a proxy issued by the Company in relation to the meeting,
    - (E) be received at that address (subject to any conditions or limitations specified in the notice) at any time before the time for holding the meeting or adjourned meeting at which the person named in the appointment proposes to vote; or
  - (ii) in the case of a poll taken after the date of the meeting or adjourned meeting, be deposited or received as aforesaid at any time before the time appointed for the taking of the poll,

and a Proxy Notice which is not deposited, delivered or received in a manner so permitted shall be invalid. Any valid Proxy Notice shall, unless stated to the contrary in it, be valid both for the relevant meeting and for any adjournment of that meeting. In this article 22.6, "address" includes a number or address used for the purposes of sending or receiving documents or information by electronic means.

- (b) A person who is entitled to attend, speak or vote (either on a show of hands or on a poll) at a general meeting remains so entitled in respect of that meeting or any adjournment of it, even though a valid Proxy Notice has been delivered to the Company by or on behalf of that person.
- (c) An appointment under a Proxy Notice may be revoked by delivering to the Company at the registered office or at any other place or address specified by the Company pursuant to article 22.6(a) in relation to the delivery of Proxy Notices for the particular meeting concerned, a notice of revocation of authority in writing given by or on behalf of the person by whom or on whose behalf the Proxy Notice was given.
- (d) A notice revoking a proxy appointment or the authority of a person authorised by a corporation pursuant to section 323(1) Companies Act 2006, only takes effect if it is delivered before the start of the meeting or adjourned meeting to which it relates or (in the case of a poll taken after the date of the meeting or adjourned meeting at which the poll was demanded) before the time appointed for taking the poll to which it relates.
- (e) Subject to article 22.6(d), the provisions of sections 330(1) to (4) inclusive Companies Act 2006 shall apply mutatis mutandis to any termination of the authority of a person authorised by a corporation pursuant to section 323(1) Companies Act 2006.
- (f) If a Proxy Notice is not executed by the person appointing the proxy, it must be accompanied by written evidence of the authority of the person who executed it to execute it on the appointor's behalf.
- (g) Without limiting the foregoing, in relation to any uncertificated shares, the directors may from time to time:
  - (i) permit appointments of a proxy by means of a communication sent in electronic form in the form of an uncertificated proxy instruction; and
  - (ii) permit supplements to, or amendments or revocations of, any such uncertificated proxy instruction by the same means.

The directors may in addition prescribe the method of determining the time at which any such uncertificated proxy instruction is to be treated as received by the Company or a participant acting on its behalf. The directors may treat any such uncertificated proxy instruction which purports to be or is expressed to be sent on behalf of a holder of a share as sufficient evidence of the authority of the person sending that instruction to send it on behalf of that holder.

#### 23. COMPANY SECRETARY

The Company shall not be required to have a secretary, but may choose to have one. Any Company Secretary shall be appointed by the directors for such term and at such remuneration and upon such conditions as they think fit, and any Company Secretary so appointed may be removed by them.

## 24. **AUTHENTICATION**

Any director or the Company Secretary or any person appointed by the directors for the purpose shall have power to authenticate any documents affecting the constitution of the Company, any resolutions of the Company or decisions of the directors or any committee of the directors, and any books, records, documents and accounts relating to the business of the Company, and to certify copies of, or extracts from, them as true copies or extracts. A document purporting to be a copy of a resolution, or an extract from the minutes of a meeting, of the Company or of the directors or any committee of the directors which is certified in accordance with this article shall be conclusive evidence in favour of a person dealing with the Company upon the faith of that document that such resolution has been duly passed or, as the case may be, that such minutes or extract is a true and accurate record of proceedings at a duly constituted meeting.

#### 25. COMPANY SEALS

- 25.1 Any common seal may only be used by the authority of the directors and the directors may decide by what means and in what form any common seal is to be used.
- 25.2 Unless otherwise decided by the directors, if the Company has a common seal and it is affixed to a document, the document must also be signed by at least one authorised person in the presence of a witness who attests the signature. For the purposes of this article, an authorised person is:
  - (a) any director of the Company;
  - (b) the Company Secretary; or
  - (c) any person authorised by the directors for the purpose of signing documents to which the common seal is applied.

## 26. PROVISION FOR EMPLOYEES ON THE CESSATION OF BUSINESS

The directors may decide to make provision for the benefit of persons employed or formerly employed by the Company or any of its subsidiaries (other than a director or former director or shadow director) in connection with the cessation or transfer to any person of the whole or part of the undertaking of the company or that subsidiary.

#### 27. NOTICES AND COMMUNICATIONS

27.1 Except as otherwise provided in these articles and subject to article 27.4, any document or information to be given, sent or supplied under these articles by the Company shall be given, sent or supplied in any way in which the Company may send or supply documents or information generally to the intended recipient under schedule 5 of the Companies Act 2006

(which may include, without limitation, in hard copy form, in electronic form or by making it available on a website) subject to, and in accordance with, the requirements of that schedule.

- 27.2 Except as otherwise provided in these articles and subject to article 27.4, any document or information to be given, sent or supplied under these articles to the Company shall be given, sent or supplied in English and otherwise in any way in which documents or information generally may be sent or supplied by the sender to the Company under schedule 5 of the Companies Act 2006 (where the sender is a body corporate) or schedule 4 of the Companies Act 2006 (in all other cases) subject to, and in accordance with, the requirements of schedule 4 or schedule 5 of the Companies Act 2006, as applicable.
- 27.3 Articles 27.1 and 27.2 shall apply whether the document or information is authorised or required to be sent or supplied by the Companies Acts or otherwise. References in this article 27 to documents or information being given, sent or supplied by or to the Company include references to documents or information being given, sent or supplied by or to the directors of the Company acting on the Company's behalf.
- 27.4 Articles 27.1 and 27.2 shall apply as if schedules 4 and 5 of the Companies Act 2006 required documents and information sent by post to be sent by prepaid first class post or (in the case of a registered address outside the United Kingdom) by prepaid airmail.
- 27.5 The Company may send or supply documents or information to or on a member through a relevant system, where the document or information relates to uncertificated shares.
- 27.6 In the case of joint shareholders of a share, all notices, documents and information shall be given to the joint shareholder whose name stands first in the register of members in respect of the joint shareholding and any notices, documents and information so given shall be sufficiently given to all the joint shareholders. A shareholder whose registered address is not within the United Kingdom and who gives to the Company an address within the United Kingdom at which notices, documents or information may be given to him, or an address to which notices, documents or information may be sent by electronic means, shall be entitled to have such notices, documents or information given to him or her at that address.
- 27.7 In the case of the death or bankruptcy of a shareholder, the Company shall not be obliged to send any documents or information to an address provided to the Company by the Transmittee(s) of such shareholder unless such Transmittee(s) has also provided the directors with such evidence of the entitlement of the Transmittee(s) to those shares as the directors shall in their absolute discretion require. Nothing in this article shall require the directors to investigate the entitlement of any person claiming to be a Transmittee of a shareholder.
- 27.8 Proof that an envelope containing a notice, document or information was properly addressed, prepaid first class and posted shall be conclusive evidence that the notice, document or information was given. Proof that a notice, document or information sent by electronic means was sent or given in accordance with guidance issued by the Institute of Chartered Secretaries and Administrators shall be conclusive evidence that the notice, document or information was sent or given. The board may require authentication of any document or information given, sent or supplied to the Company in electronic form in such manner as it may determine.

- 27.9 Section 1147 of the Companies Act 2006 shall not apply to documents or information sent by or to the Company for the purposes of the Companies Acts or these articles.
- 27.10 In this article 27, "address" includes a number or address used for the purposes of sending or receiving documents or information by electronic means.
- 27.11 Nothing in these articles shall affect any legal requirement that any particular notice or other document be served in any particular manner.

## 28. INDEMNITIES AND FUNDING OF PROCEEDINGS

- 28.1 Subject to the provisions of and so far as may be consistent with the Companies Act 2006:
  - (a) the directors may exercise all the powers of the Company to indemnify any person who is, or was at any time a director of the Company or any of its associated companies, against all liabilities incurred by or attaching to him or her in connection with his or hers duties, powers or office in relation to any such company of which he or she is or was a director, to the fullest extent permitted by law;
  - (b) where the Company or any of its associated companies is or was at any time a trustee of an occupational pension scheme (as defined in section 235(6) of the Companies Act 2006 as amended, modified or re-enacted from time to time), the directors may exercise all the powers of the Company to indemnify any person who is or was at any time a director of that company against all liabilities incurred by him or her in connection with that company's activities as trustee of the occupational pension scheme, to the fullest extent permitted by law; and
  - (c) the directors may exercise all the powers of the Company to provide any director of the Company or of its holding company with funds to meet expenditure incurred or to be incurred by him or her of the kind referred to in sections 205(1)(a) and 206(a) of the Companies Act 2006 as amended, modified or re-enacted from time to time and otherwise take any action to enable any such director to avoid incurring such expenditure, to the fullest extent permitted by law;

and in this article 28.1 the term **"associated company"** shall have the meaning given in section 256(b) Companies Act 2006 as amended, modified or re-enacted from time to time.

# 29. INSURANCE

- 29.1 Without prejudice to article 28, the directors may exercise all the powers of the Company to purchase and maintain insurance for, or for the benefit of, any person who is or was at any time:
  - (a) a director of any Relevant Company; or
  - (b) a trustee of any pension fund or retirement, death or disability scheme for the benefit of any employee of any Relevant Company or of any employees' share scheme in which employees of any such Relevant Company are interested;

including (without limitation) insurance against any liability referred to in article 28 attaching to him or her in relation to any Relevant Company, or any such pension fund, retirement or other scheme or employees' share scheme.

- 29.2 In article 29.1, "**Relevant Company**" means the Company or any other undertaking which is or was at any time:
  - (a) the holding company of the Company; or
  - (b) a subsidiary of the Company or of such holding company; or
  - (c) a company in which the Company has an interest (whether direct or indirect).

#### 30. DATA PROTECTION

Each of the members and directors consent to the processing of their personal data by the Company, the members and directors (each a "**Recipient**") for the purpose of due diligence exercises, compliance with applicable laws, regulations and procedures and the exchange of information among themselves. A Recipient may process the personal data either electronically or manually. The personal data which may be processed under this article shall include any information which may have a bearing on the prudence or commercial merits of investing, or disposing of any shares (or other investment or security) in the Company. Other than as required by law, court order or other regulatory authority, that personal data may not be disclosed by a Recipient or any other person except to a Group Companies and funds managed by any of the Group Companies. Each of the members and directors consent to the transfer of relevant personal data to persons acting on behalf of the Recipient and to the offices of any Recipient both within and outside the European Economic Area for the purposes stated above, where it is necessary or desirable to do so.